CITY OF ST. PETE BEACH



PERSONNEL RULES AND REGULATIONS

Effective: May 12, 2020 Update 10/27/20

TABLE OF CONTENTS

		Page
Section 1*	General Provisions	1
1.01	Purpose	1
1.02	Positions Covered	1
1.03	Administration	2
1.04	Overall Employment Policy	
1.05	Collective bargaining agreement (CBA)	3
1.06	Amendments	3
1.07	Department Policies	3
1.08	Memos, Policies and Regulations	3
SECTION 2*	Definition of Terms	4
SECTION 3*	Standards of Conduct	9
3.01	Policy	9
3.02	Conflict of Interest	9
3.03	Political Activity	10
3.04	Employment of Relatives	
3.05	Outside Employment	
3.06	Release of Information	
3.07	Solicitation and Distribution	
3.08	Employees' Personal Conduct	
3.09	Bribery	
3.10	Acceptance of Gifts, Gratuities, and/or Benefits	14
3.11	Additional Ethical Conduct Requirements For Employees With	4-
0.40	Regulatory Responsibilities	17
3.12	Additional Ethical Conduct Requirements For Reporting	40
0.40	Individuals And Procurement Employees	
3.13	Whistleblower Policy	
3.14	Use of City Property	
3.15	Uniforms, Dress and Appearance	
3.16	General Prohibitions	25
SECTION 4*	Types and Terms of Employment	26
4.01	Basis of Employment	
4.02	Participation in Benefits	
4.03	Employment Terms	
4.04	Regular Full-Time At-Will Employees – Special Conditions	
4.05	Probationary Employees	
4.06	Promotional Probation	
4.07	Loyalty Oath	28

SECTION 5*	Hiring Procedures	29
5.01	Vacancies	29
5.02	Interviews	
5.03	Basis for Selection	30
5.04	Testing	30
5.05	Drug Testing	
5.06	Disability and Medical Examinations	
5.07	Pay Rates for New Employees	
SECTION 6*	Types of Separations	32
6.01	Types of Separations	
6.01(A)	Death	
6.01(B)	Disability	
6.01(C)	Dismissal or Discharge	
6.01(D)	Reduction in Force (Layoff)	32
6.01(E)	Probationary Termination	
6.01(F)	Resignation	32
6.01(G)	Retirement	33
6.02	Return of Property and Financial Obligations	33
		_
SECTION 7	Transfers and Work Out of Classification	
7.01	Temporary Transfers/Work Out of Classification	
7.02	Permanent Transfers	
7.03	Reclassification	34
SECTION 8	Promotions/Demotions/Laterals	36
8.01	Notice of Job Vacancy (or Posting)	
8.02	Application	
8.03	Pool of Qualified Applicants	
8.04	Interview	
8.05	No Sufficiently Qualified Applicants	
8.06	Basis of Selection	
8.07	Wage Rate	
8.08	Promotion, Demotion & Lateral	
0.00	Tromotion, Demotion & Lateral	
SECTION 9	Seniority Layoff and Recall	38
9.01	Accrual	
9.02	Loss of Seniority	
9.03	Layoff Selection	
9.04	Permanent Layoffs	
9.05	Recall	
9.06	Decisions Final	
3.00		
SECTION 10*	Attendance/Tardiness	4 1
10.01	Present and On Time	41

10.02	Call-In	4 1
10.03	Verification	
10.04	Continuing Absence	41
10.05	Person to Call	41
SECTION 11	Disciplinary Action	42
11.01	General Statement	
11.02	Forms of Discipline	
11.03	Application of Disciplinary Action	
11.04	Notice Of Disciplinary Action And Pre-Disciplinary Procedure	
11.05	Types of Offenses	
11.06	Suspension Pending Resolution of Criminal Charges	47
SECTION 12	Appeal Procedure For Disciplinary Actions/Name Clearing	
	Hearings For Terminated Employees	49
12.01	Written Warnings	49
12.02	Suspensions Without Pay, Demotion And Termination Appeal	
	Procedure	49
12.03	General Provisions	49
12.04	Delegation of Authority	
12.05*	Name Clearing Hearing	50
SECTION 13	Employee Grievance Procedure – Written Warnings and	
	Non-Disciplinary Matters	52
13.01	Purpose	52
13.02	Definition of a Grievance	
13.03	Procedure	52
SECTION 14	Hours of Work and Overtime	54
14.01*	Hours of Work	54
14.02	Regular WorkWeek or Period	54
14.03*	Overtime	
14.04	Assignment and Working Overtime	
14.05	Hours Counted	
14.06	Call Backs	
14.07	Telecommuting	56
SECTION 15	Holidays	57
15.01	List of Legal Holidays/Days Observed	57
15.02	Requirements To Receive Holiday Compensation	58
15.03	Absence Due To Sickness	58
15.04	Twenty-Four Hour Shift Employees	59

Final iv

Section 16	Paid Time Off (PTO) Policy	60
16.01	Definition	60
16.02	Eligibility And Usage	60
16.03	Accrual	61
16.04	Approval	61
16.05	Department Directors	62
16.06	Charging Time	62
16.07	Unused PTO Time	62
16.08	Payment Of Unused PTO	
16.09	Unused Sick Leave Balance as of August 19, 2013	63
Section 17	Bereavement Leave	
17.01	Bereavement Leave	65
Section 18*	Family Medical Leave Policy	
18.01	Eligibility And Reasons	
18.02	Definitions	
18.03	Measuring the Twelve Monty Period and Counting FMLA Leave	
18.04	Intermittent Leave or Leave on a Reduced Schedule Basis	
18.05	Employee Notice and Certification Requirements	
18.06	Health Insurance Premiums	
18.07	Accrual	
18.08	Substitution Of Accrued Paid Leave	
18.09	Job Restoration Upon Return From Fmla Leave	
18.10	Failure To Cooperate	71
Section 19	Court/Witness Leave	
19.01*	Witness Leave For The City	
19.02	Other Court-Related Leave	
19.03	Jury Duty	
19.04	Return To Work	
19.05	Reporting	73
Section 20	Leave of Absence Without Pay	74
20.01	Leave Of Absence Without Pay	74
Section 21	Effect Of Leaves On Insurance Coverage	
21.01	Effect Of Leaves On Insurance Coverage	76
Section 22*	Military Leave	77
Section 23*	Prohibition Of Harassment Policy	78
23.01	Policy	

23.02	Definitions and Exampleas	78
23.03	The Prevention of Harassment in the Workplace is Every	
	Employees Responsibility	79
23.04	Complaint and Investiagation Procedure for Prohibited	
	Harassment	80
23.05	Internal Investigation Procedure	81
0	D	0.0
Section 24*	Pension	
24.01	Section 457 Deferred Compensation Plan	
24.02	(Local Ordinance) Pension Plan	
24.03	Details And Copies	83
Section 25	Performance Evaluations	84
25.01	The Performance Evaluation System	
25.02	Use	
25.03	Review	
25.04	Maintenance Of Records	
Section 26*	Drug-Free Workplace and Alcohol Policy	85
26.01	Policy	
26.02	Prohibitions	
26.03	Use Of Legal Drugs	
26.04	Testing	86
26.05	Reporting and Conviction of Alleged Crimes Including Drugs or	
	Alcohol	
26.06	Discipline For Violation of Policy	
26.07	Employee Injured on the Job	
26.08	Employee Assistance Program	89
26.09	Reporting Violation of the Policy	90
26.10	Coordination With The Human Resources Office	91
Section 27	Educational Reimbursement Program And Incentive Pay	92
27.01	Tuition	
27.01	Eligibility	
27.03	Requests And Payment	
27.04	Repayment By The Employee	
27.05	Schedules	
27.05 27.06	Incentive Pay	
27.00	incentive Fay	93
Section 28*	Miscellaneous	94
28.01	Smoking/Tobacco Products	94
28.02	Unemployment/Re-Employment Compensation	95
28.03	Workers' Compensation	
28.04	Light Duty	

Final vi

28.05	Desks, Lockers, Other City Property/Equipment, Personal	
	Property	96
28.06	Records and Reports	
28.07	Employee Training and Development	
28.08	Deductions	
28.09	Cell Phones	98
28.10	Electronic Communications, Including Text Messages, Usage And Retention	99
28.11	Social Media Policy - Employees	.100
28.12	Take-Home Vehicles	
28.13	Safety	.104
Section 29*	Group Insurances	.106
29.01	Health and Dental Insurance	
29.02	Dependent Coverage	.106
29.03	Life Insurance	.106
29.04	Retirees	.106
29.05	Carousel Benefits Program	.106
29.06	Eligibility	
29.07	Reservation of Rights	

Final vii

WELCOME TO THE CITY OF ST. PETE BEACH

The City of St. Pete Beach, Florida welcomes you as a member of its municipal staff. As a City employee, you are an important part of a group of select individuals dedicated to providing the most efficient, professional and courteous service possible for our City's customers - its citizens, property owners and visitors. We hope your association with us will be a happy and successful one. While our resources are directed toward serving our community, we are vitally concerned about your well-being, as an employee. An organization is defined by its values -- the principles or standards that the organization considers to be important. Recognizing that the public is our ultimate customer, the employees of St. Pete Beach exist to serve our citizenry including the City Commissioners, the press, and fellow employees. The employees of St. Pete Beach are the organization's most valuable resource in terms of service to the public, and therefore the City places high priority on the well-being of its employees. It is through your efforts, combined with those of other employees, that the most efficient, effective and professional services can be provided. Every effort will be made to assist you in adjusting to your job and in finding satisfaction in your work.

These Personnel Rules and Regulations have been developed to establish a consistent and equitable basis for administering matters affecting the working environment in which the City's staff functions. The policies contained in this manual are administered by the City Manager. Any questions you may have regarding this manual can be answered by your immediate Supervisor or the Human Resources (HR) Administrator. These policies are not to be construed as a contract of employment for any definite period of time between you and the City or to vest entitlement in you to any of the benefits contained herein. For those of our employees represented by a union, to the extent the collective bargaining agreement conflicts with these policies or the benefits provided in these policies, the collective bargaining agreement will prevail. In all other cases the adopted rules and regulations shall apply. The Personnel Rules and Regulations contained herein supersede and replace any previous versions and any prior Employee Handbook, memoranda or materials provided regarding the subjects herein. Inasmuch as the rules and regulations contained herein are guides only, the City, subject to applicable law, reserves the right to amend, terminate, or modify any of the policies, practices and procedures contained herein. Employees will be advised of changes, deletions or additions.

We hope these Personnel Rules and Regulations answer most of your questions regarding the City's personnel/human resource practices. Should you have suggestions for improvement to these Personnel Rules and Regulations, please feel free to forward your comments to the HR Administrator or me.

Both the City Commission and other staff personnel are pleased to have you as a member of the City's team and join with me in the hope that you will enjoy both your work and the new friends you may make while associated with our City.

Alex Rey City Manager

Final viii

SECTION 1*

GENERAL PROVISIONS

1.01 PURPOSE

The purpose of these Personnel Rules and Regulations (hereafter referred to as "PRR" or "Rules") is to establish procedures, which will serve as a guide to administrative actions covering most personnel actions, which may arise. The final interpretation and application of these Rules shall be made by the City Manager. The City reserves the right to amend, alter, modify, delete and add to these Rules.

1.02 POSITIONS COVERED

- A. Unless a specific Section or Subsection provides otherwise, the provisions of these Rules shall be applicable to all employees in City service except:
 - Elected officials.
 - 2. Persons hired as independent contractors on a contractual, fee, or retainer basis (e.g. City Attorney)
 - 3. Temporary or probationary employees.
 - 4. Persons employed under the provisions of government programs or grants unless they are classified as regular full-time employees by the City.

Provided, however, the sections or subsections with an asterisk (*) beside them apply to all employees.

- B. Employees employed as the City Manager's Executive Assistant, Department Directors, Assistant Department Directors, Department Operations Managers, City Engineer, and the HR Administrator are classified "at will" and:
 - Shall be supervised and evaluated by the City Manager and shall serve in their capacity - and as City employees at his will and pleasure.
 - 2. Shall not have access to Sections 12 or 13 but shall present their grievances directly to the City Manager who shall resolve same as he deems in the best interest of City operations.

- C. In addition to subparagraph (B) above, employees in positions in pay grades 122 or higher serve in their position at the will and pleasure of the City Manager.
- D. The City Clerk shall serve at the will and pleasure of the City Commission and shall have day-to-day supervision of all aspects of the City Clerk's office. Employees of the City Clerk's office shall be hired, dismissed, supervised and evaluated solely by the City Clerk, or his designee, and shall have the duties as prescribed, exclusively, by the City Clerk.

1.03 ADMINISTRATION

- A. General Administration: The City Manager or HR Administrator shall be responsible for the City's personnel administration.
- B. Directors are responsible for the efficient and effective operation and the direct supervision of the employees assigned to their department or area of responsibility and for the proper and effective administration and enforcement of these Rules. These duties may be delegated by the Director, but the ultimate responsibility shall remain with the Director.

1.04 OVERALL EMPLOYMENT POLICY

The overall employment policy of the City shall include:

- A. There shall be no illegal discrimination in employment, employment opportunities or job actions on the basis of race, color, religion, age, sex, national origin, disability, marital status, veteran status, sexual orientation or other legally protected category. No job applicant or present employee will be illegally discriminated against or given preference because of any of the above characteristics, unless otherwise required by law.
- B. Persons with known legally recognized disabilities will be given full consideration for employment and opportunities for advancement in all departments and divisions. In accordance with the law, the City will make reasonable accommodations with respect to the essential functions of the job (as well as during the application process), provided the person is otherwise qualified to perform the job, and provided further such accommodation does not create undue hardship on City operations. To request a reasonable accommodation under the ADA or other applicable law, please see HR. You may be required to complete forms and/or provide a medical statement from a doctor.
- C. The City will make all other legally required accommodations including religious accommodations. If you believe you need an accommodation due to your sincerely held religious beliefs, please see HR.

D. The City will take affirmative recruitment actions to expand employment opportunities for groups that are underutilized in the City workforce, but not in any way which violates applicable law.

1.05 COLLECTIVE BARGAINING AGREEMENT (CBA)

Where these Rules or departmental rules and regulations are in conflict with the express terms of any CBA, the terms of the CBA shall take precedence.

1.06 AMENDMENTS

Amendments to the PRR shall be promulgated by the City Manager. Copies will be distributed to all Departments and employees for review.

1.07 DEPARTMENT POLICIES

- A. Departmental policies and standard operating procedures related to personnel management will be in writing and submitted to and reviewed by the City Manager for approval.
- B. Departmental policies and standard operating procedures related to personnel management approved by the City Manager will serve as supplements to these Policies. In the event of conflict, the PRR shall prevail unless the departmental rule has been specifically approved as an exception by the City Manager.
- C. Approved changes in departmental policies and standard operating procedures related to personnel management shall be distributed to the affected employees after approval.

1.08 MEMOS, POLICIES AND REGULATIONS

All prior memos, policies, procedures and regulations inconsistent with this PRR are null and void.

SECTION 2*

DEFINITION OF TERMS

For the purposes of the Personnel Rules and Regulations, the following terms and related definitions shall apply and shall be interpreted and applied by the City, or its designee.

Active Pay Status - When an employee is working, on authorized paid time off, paid holidays or other time where pay is being credited to employee.

ADA - Americans with Disabilities Act

Anniversary Date – The original date an employee begins employment and the same date in following years. This is the date upon which entitlement to fringe benefits are based unless a specific benefit provides otherwise. The anniversary date may be changed in accordance with these Rules.

At-Will Employee – Employees who serve at the pleasure of the City, or its designee, may be removed from their position and disciplined up to and including termination for any or no reason subject only to applicable law. Refer to Sections 1.02(B), (C), and (D) and 4.03 for a listing of at-will positions.

Benefits - indirect, non-cash, or cash compensation paid to qualified employees above and beyond regular salary or wages, including but not limited to PTO, other types of leave, health insurance, life insurance, dental insurance, vision insurance, employee assistance program, retirement/pension plans, HSAs, FSAs, HRAs, short term disability insurance, and tuition reimbursement.

CBA – A collective bargaining agreement.

Calendar Year - For the purposes of recording leave, such as emergency, personal, military and all other leaves that have a calendar year limitation (excluding FMLA), the dates used for reporting W-2 wages for employees shall be considered the calendar year.

City – City Commission of the City of St. Pete Beach or an employee authorized to act on behalf of the City with respect to a particular matter.

City Manager - The City Manager, or its designee.

City Seniority - The total time an employee has continuously worked for the City without loss of seniority under Section 9.

Classification Plans – Commission approved pay structures.

Classification Seniority (also referred to as job or position seniority) - The length of time an employee has been continuously employed in his current position classification. Classification seniority will be lost or changed upon the loss of seniority as described under Section 9.02, and/or the transfer, promotion, demotion or reassignment to or from one job classification to another.

Collective Bargaining Agreement (also referred to as "CBA") - An agreement between an employee organization and the Commission negotiated and ratified as required by the Public Employees Relations Act.

Commission - The City Commission of the City of St. Pete Beach.

Compensatory Time (also referred to as Comp Time) - Time off from work in lieu of pay.

Confidential Employee - An employee who assists a managerial employee in a confidential manner relating to labor relations and is exempt from coverage of the Public Employee Relations Act.

Demotion - Permanent reassignment of an employee to a lower level job classification for disciplinary reasons. Disciplinary demotions are covered under Section 8.08 and Section 11. Non-disciplinary demotions are made at will by the City Manager for operational reasons.

Departmental Seniority - The length of time an employee has been continuously employed in a department.

Department Director – A Director who is assigned the overall responsibility for the operation of a recognized department.

Designated Workweek- Midnight Sunday to 11:59 p.m. the following Saturday

Dismissal or Termination - Involuntary separation from City employment.

DOL – Department of Labor.

Domestic Partner - Domestic Partner shall mean a person who is neither married to nor related by blood or marriage to the employee; is the employee's sole spousal equivalent; lives together with the employee in the same residence and the employee and Domestic Partner intend to do so indefinitely; is responsible with the employee for each other's welfare. The employee may not be married to another person at the time of declaration of domestic partnership; or have declared a domestic partnership within the prior twelve (12) months. In order to declare domestic partnership, the employee must execute a Declaration of Domestic Partnership.

DOT – Department of Transportation.

Driving Position - A position where the employee drives or may be required to drive a City vehicle in the performance of his duties.

Employee -

- A. Regular full-time employee is any employee who is non-probationary and assigned a regular schedule of a minimum of forty (40) hours a week or the regular Section 7(K) schedule (Fire), whichever applies.
- B. Regular Part-time employee is any employee who is assigned a regular schedule of less than forty (40) hours per week.
- C. A probationary employee is any full-time employee who has not completed the initial probationary period.
- D. Temporary is any employee hired for a fixed period of time.

Exempt Employee - An employee exempt from the minimum wage and/or overtime under the Fair Labor Standards Act and paid a salary for all hours worked in a work week.

FCRA - The Florida Civil Rights Act

FLSA - The Fair Labor Standards Act

FMLA - Family Medical Leave Act

Flexible Scheduling – A method of scheduling typically used for adjusting an hourly employee's work hours (start and stop times) within an eight (8) hour workday or forty (40) hour workweek.

He/His/Him - Are generic and used for reference purposes only to signal reference to both males and females.

Immediate Family - Includes spouse, domestic partner, children, parent, brother, sister, In-laws (father, mother, brother, sister, son or daughter only), any relative living in the same household, stepparent, stepchild, step brother or sister, grandmother, grandfather and grandchild, legal guardian, provided such definition shall not apply to FMLA leave under Section 18.

Job Description - A written description of some but not all of the duties and responsibilities of a job.

Managerial/Confidential Employee - An employee who performs jobs that are not routine and clerical in nature, and which require the exercise of independent judgment and is exempt from coverage of the Public Employee Relations Act.

MAPS Classification Plan – (Managerial, Administrative, Professional, Supervisory) payfor-performance pay plan based on established goals.

May - The word "may" shall be interpreted as permissive.

Merit Pay Increase - An increase in compensation established in the Classification Plan which may be granted to an employee based on merit.

Nonexempt Employees – Employees that are entitled to overtime pay according to the Fair Labor Standards Act (FLSA).

Paid Time Off (PTO) – PTO is an all-inclusive flexible time off policy which takes the place of traditional individual vacation, sick, and injury leave programs.

Pay Grade - The salary range, which is assigned to a particular classification title, expressed as a pay range number.

Performance Evaluation (also referred to as "PE") - A written report of an employee's job performance.

Permanent Transfer - The reassignment of an employee from one position to another as provided in Section 7.02.

Probationary Period - The first 2080 hours, or 2912 hours for twenty-four hour shift employees, of continuous employment as a regular full or part-time employee upon hiring or promotion; provided where a formal training period, certification or license is required for a job, the probationary period shall continue until the certification or license is obtained. After successful completion of the probationary period, the employee will be classified as a regular employee.

Promotion - Subject to completion of the promotional probationary period of 2080 hours, or 2912 hours for twenty-four hour shift employees, regular assignment of an employee to a higher-level job classification.

Reclassification - Movement of a job or job classification from one pay grade to another based on significant changes in the job duties, responsibilities, job market and/or other work-related factors.

Reemployment - The hiring of a person who formerly worked for the City. Persons rehired shall be new employees for all purposes, unless the Director recommends, and the City Manager approves otherwise in a particular case.

Reinstatement - An action returning a person to City employment.

Resignation - Act of voluntarily withdrawing from City employment.

Section 7(K) - Section 7(K) of the Fair Labor Standards Act allowing certain employees in the Fire Department to be paid overtime based on a schedule other than after forty (40) hours in a seven (7) day period.

Shall - The word "shall" will be interpreted as mandatory.

Status Date – The day an employee was hired, permanently transferred, promoted or demoted to his current position. Also changing from Part-time to Full-time, from temporary to regular employment or vice versa.

Workday - The scheduled number of hours an employee is required to work per day.

Work Week or Work Period - The number of hours regularly scheduled to be worked during any seven (7) consecutive days or other work period allowed by the Fair Labor Standards Act and is authorized by the City.

SECTION 3*

STANDARDS OF CONDUCT

3.01 POLICY

- A. One of the primary objectives of the City of St. Pete Beach, in accordance with the Charter, is to establish and administer a system of personnel administration consistent with the goal of providing superior service to the community by employing and retaining individuals of the highest caliber who display pride and dignity in the performance of their duties in a public service career.
- B. To an unusual extent and in a special way, employees in the City organization are "Good Will Ambassadors". Such status involves a degree of duty and obligation regarding public and private conduct which is not common to other classes of employment. The attitude and deportment of a City employee should at all times be such as to promote the good will and favorable attitude of the public toward the City, its programs, and policies.
- C. All employees are encouraged to develop skills and seek formal training that will enhance their personal development and add to the overall expertise of the organization.
- D. It is the policy of the City to expect from employees compliance with this PRR, State Statutes, Federal Regulations and Departmental Rules in the performance of duties, as well as compliance with all safety rules and standards. An employee who violates any of the Standards of Conduct, Departmental Rules, or the PRR shall be subject to disciplinary action.

3.02 CONFLICT OF INTEREST

- A. Employees in a position to influence actions and decisions of the City or a member of the managerial staff shall refrain from internal or external relationships which may adversely affect the exercise of their independent judgment in dealing with suppliers of goods or services and other persons not employed by the City.
- B. No City employee shall accept any gift, loan, favor or service that might reasonably appear to improperly influence them in the discharge of their official duties.
- C. An employee shall not use his position with the City to obtain or attempt to obtain any special preferences, favors, privileges or exemptions for himself or for any other person.

- D. No employee shall disclose confidential information gained by reason of his official position with the City except in and as a part of his normal duties as a City employee; nor shall such employee use such confidential information not available to the public for personal gain or benefit.
- E. When an employee has or anticipates creating a business relationship with another person, partnership, firm, corporation or other business entity which does or seeks to do business with the City, or any division thereof, the employee shall advise the City Manager in writing as soon as that potential relationship is known. Failure to so advise the City Manager may result in immediate termination. The City Manager will determine whether there is a conflict of interest or a potential conflict of interest and direct the employee's activities in such a way that the conflict or potential conflict of interest no longer exists. The City Manager's determination as to whether there is a conflict of interest or a potential conflict of interest and the actions required by the City Manager shall be final.
- F. No City employee shall transact any business in their official capacity with any business entity of which he or she is an officer, Director, agent, or member, or in which he or she owns a controlling interest.
- G. No employee shall have or hold any employment or contractual relationship with any business entity or any agency that is subject to the regulation of or is doing business with the City, excluding those organizations and their officers who, when acting in their official capacity, enter into or negotiate a collective bargaining contract with the City; nor shall an officer or employee of the City have or hold any employment or contractual relationship that will create a continuing or frequently recurring conflict between his private interests and the performance of his public duties or that would impede the full and faithful discharge of his public duties. This shall not prohibit an employee from practicing in a particular profession or occupation when such practice is required or permitted by law or ordinance.
- H. All employees shall comply with the Code of Ethics for Public Officers and employees under Florida State Statutes 112, as well as all City ordinances and policies, including this PRR.
- I. An employee shall not use his employment with the City to attempt to persuade any person, including, but not limited to, citizens, residents or guests, to make contributions to any cause, unless that cause has been specifically approved by the City Manager.

3.03 POLITICAL ACTIVITY

A. Employees may engage in political activities during their non-duty time so long as their activities do not interfere with the operation of City business.

- B. Employees shall not wear or display political badges, buttons or stickers when on duty, when wearing a City insignia, riding in or on City equipment or when in a City uniform.
- C. Employees shall comply with all state and local laws involving political activity.
- D. Employees may run for elective office or be appointed to non-elective office other than those involving the City of St. Pete Beach so long as the position in no way interferes with their work as a City employee.
- E. Subject to applicable law the restrictions set forth in that law, employees may run for City office so long as they handle their candidacy in such a manner as to not interfere with the efficient operation of the City.
- F. No City employee shall solicit any contribution for the campaign fund of any candidate for City office or take part in the political campaign of any candidate for City office while on duty.
- G. No City employee shall solicit, orally or by letter, or be in any other manner involved in obtaining any assessments, contributions or services for any political candidate or party from any employee while on duty.

3.04 EMPLOYMENT OF RELATIVES

- A. The term relative includes the following relationships: Immediate Family as defined in "Definition of Terms" and uncle, aunt, first cousin, nephew, niece. Under no circumstance shall an employee have supervisory authority (direct/or indirect) over a relative or be employed in a position that creates or there is a likelihood may create a conflict of interest. In some areas or departments, the City Manager may determine if it is in the best interest of the City to ensure discipline and operational harmony that relatives not be hired or employed.
- B. Current employees who marry or otherwise acquire the relationship of relative with another employee during the term of employment shall notify the City Manager/HR Administrator. In the event it is determined the relationship is inconsistent with subparagraph a, the City Manager/HR Administrator shall make an effort to reassign one (1) of the employees so as not to have a conflict with this policy. The decision by which an employee is reassigned should be based on the qualifications of the employee and the availability of suitable positions within another department. If reassignment is not acceptable or available for resolving the conflict, one (1) of the employees shall be separated from employment. If the employees/relatives involved cannot agree as to which person is to terminate their employment, the decision shall be made at the discretion of the City Manager based on the needs of the City.

Employees/relatives who would be disqualified from employment as a result of this directive, but were employed under such status as of the effective date of this section, are grandfathered in for their status as of that date and will not be required to terminate their employment, although such employees may be subject to transfer to minimize any potential for conflict. Further, such employees will not be eligible for promotions or transfers if such would result in the creation of a conflicting situation as described in this policy. Employees/relatives who were employed by the City prior to the effective date of this rule and who would otherwise thus be entitled to continue their employment with the City, but who terminate employment after the effective date of this policy, shall be subject to this rule with regard to any future employment or reemployment.

No employee who has the ability to appoint, employ, promote, affect compensation, advance employees or to recommend such action shall do so with regard to a relative.

3.05 OUTSIDE EMPLOYMENT

- A. Employees will not engage in outside employment, except National Guard, Reserves or similar military duty, which may in any way hinder the proper performance of their public duties or impair the efficiency of City employees as determined by the Employer.
- B. No employee shall engage in outside employment with, or render services for, any person or business transacting business with any agency or department of the Employer without approval of the Department Director and the City Manager.
- C. Employees who engage in secondary employment shall do so only with approval of the Department Director and the understanding and acceptance that their primary duty and obligation and responsibility are to the Employer. Employees who engage in secondary employment will provide a source of telephone communications with the place of off-duty employment, and such information shall be current at all times.
- D. All employees are subject to call at any time for emergencies for overtime duty, and no secondary employment may interfere with this obligation. Employees may be subject to disciplinary action under this section if they fail to report for emergency or overtime duty after being ordered to do so.
- E. The City may require proof of workers' compensation coverage by the outside employer. An employee shall not drive any City owned vehicle to his outside employment, nor take any City owned equipment to said employment.
- F. Employee may not work at outside employment while on a leave of absence from the City, except with the authorization of the City Manager.

G. Employees who are injured while working another job or jobs are required to notify their Director, or his designee, immediately.

3.06 RELEASE OF INFORMATION

Employees may accept a request for information from the public but shall forward it to the City Clerk, custodian of public records, for fulfillment.

3.07 SOLICITATION AND DISTRIBUTION

- A. Employee contributions to recognized charitable organizations are purely voluntary. No coercion of an employee to make contributions shall be permitted.
- B. Employees of the City are prohibited from conducting or promoting private business for gain while on duty or during scheduled working hours of any of the employees involved or within any City building.
- C. Employees are prohibited from soliciting for any reason during time they or the person they seek to solicit are being paid to perform actual work, including solicitations in behalf of or in opposition to a labor organization under circumstances which management determines interfere with the efficient operation of the City.
- D. Employees are prohibited from distributing literature of any kind during hours they are being paid to perform actual work or in any area where employees are engaged in work at any time under circumstances which management determines interfere with the efficient operations of the City.
- E. The solicitation and distribution prohibitions set forth in paragraphs C and D above shall not apply to solicitation and/or distribution by the City or its managerial staff, when such is part of the normal operation of City business.

3.08 EMPLOYEES' PERSONAL CONDUCT

Employees shall handle their personal life, including their financial obligations, in such a manner that it will not interfere with the efficient operation of City business or the performance of their own job responsibilities.

3.09 BRIBERY

A. An attempt to bribe an employee is a flagrant attack on the integrity of the City and all of its employees. A bribery attempt occurs when an offer is made to give anything of value to an employee with the intention that the employee will do something or fail to do something relating to the performance of the employee's duties. Whether the action or inaction

- would have taken place anyway is not pertinent to whether a bribery attempt has occurred.
- B. Bribery offers are often made subtly rather than in direct fashion. Employees must be perceptive and alert to solicitations to accept gifts of money or anything else of value in return for acts or omissions involving their official functions and shall promptly report suspicious offers in order that the facts can be evaluated, and investigations initiated when required. Any employee who has reasonable grounds for believing that a bribery attempt has occurred or will occur shall:

IMMEDIATELY REPORT THE MATTER TO HIS DEPARTMENT DIRECTOR; AND COOPERATE FULLY IN ANY INVESTIGATIONS.

3.10 ACCEPTANCE OF GIFTS, GRATUITIES, AND/OR BENEFITS

- A. Employees shall not solicit or accept any gift, either directly or indirectly, from any person or entity doing business with, regulated by, or seeking to do business with the City, or from the agent or lobbyist of any such person or entity. Employees who have questions in this regard should request an advisory opinion from their Supervisor, Department Director, or the HR Administrator. Any supervisor or Department Director may, in turn, seek guidance from the City Manager and/or City Attorney.
- B. The term "gift" means any gratuity, benefit, or any other thing which is accepted by, or given to an employee or another person on the employee's behalf, either directly or indirectly, and includes by way of illustration and not limitation, the following:
 - 1. Real property and/or the use thereof;
 - 2. Tangible or intangible personal property (as defined in Section 192.001, Florida Statutes) and/or the use thereof;
 - 3. A preferential rate or terms on a debt, loan, goods, or services that is neither a government rate available to all other similarly situated government members or officials, or a rate which is available to similarly situated members of the public;
 - Forgiveness of an indebtedness;
 - 5. Transportation, lodging, or parking unless on City business and the provision of same is disclosed on a travel voucher;
 - 6. Food or beverage for employee;
 - 7. Membership dues:

- 8. Entrance fees, admission fees, or tickets to events, performances, or facilities:
- 9. Services provided by persons pursuant to a professional license or certificate;
- 10. Other personal services for which a fee is normally charged by the person providing the services; and
- 11. Any and all other similar goods or services having value not already provided for in this definition.

C. The term "gift" does not include:

- Salaries, benefits, services, fees, commissions, or expenses associated with the employee's employment with the City or outside employer;
- 2. Contributions or expenditures reported pursuant to Chapter 106, Florida Statutes; campaign-related personal services provided without compensation by volunteers or any other contribution or expenditure by a political party;
- An honorarium or an expense reimbursement related to an honorarium event paid to individuals and/or family members; however, disclosure of same shall be made, in writing, to the HR Administrator;
- 4. An award, plaque, certificate, or similar personalized item given in recognition of the employee's public, civic, charitable, or professional service;
- 5. An honorary membership in a service or fraternal organization presented merely as a courtesy by such organization;
- 6. The lawfully permitted use of a public facility or public property for an authorized and approved public purpose;
- 7. Candy, food products, or plants which are generally distributed during holidays as an act of courtesy or benevolence; however, such items shall be placed in an area which allows and encourages all employees and the public to partake of such items;
- 8. Promotional items such as office supplies, caps, t-shirts, etc. which have nominal value and are normally given away and distributed to employees;

- 9. Discounts offered to all employees (e.g., theme park discount cards) that are normally offered to other groups of employees; and
- 10. Winning a prize through a chance drawing at a conference from a vendor even when the city paid for the employee's registration and attendance.

D. Exceptions Regarding Gifts

- 1. When an employee's duties and responsibilities include attendance (such as a speaker or program participant) at business, industry or public luncheons or dinner meetings with public, industry, or business committees, organizations, or associations and these meetings are related to City business, employees may participate. When a meal is provided at the function, the employee is authorized to accept the meal.
- Employees authorized to attend meetings, seminars, conventions, etc., sponsored by professional organizations are free to participate in the social functions that are part of the meetings or sanctioned by the host.
- 3. It may be appropriate for employees to attend social functions sponsored by members of an industry, business, or profession for a broad spectrum of community groups (such as legislators, businesspersons, and local officials). However, in deciding to attend such functions, employees should be careful to avoid any appearance of impropriety.
- 4. If an organization is promoting an event, such as a concert, in the area and delivers a few tickets to the City as a courtesy, the employee who receives the tickets or invitations shall notify the HR Administrator immediately who will advise the employee as to the disposition of the gift.
- 5. Employees are prohibited from promoting private businesses while on City time and from engaging in political activity, on behalf of themselves or others while on City time.
- 6. In the event that a gift, other than for general consumption, is received by mail or "dropped off" for an individual employee, the employee shall notify the HR Administrator immediately upon becoming aware of the gift. The HR Administrator will advise the employee as to the disposition of the gift.

3.11 ADDITIONAL ETHICAL CONDUCT REQUIREMENTS FOR EMPLOYEES WITH REGULATORY RESPONSIBILITIES

The following definitions shall apply for the purposes of this Section:

- A. The term "regulatory responsibilities" means duties or responsibilities assigned to an employee which involve:
 - Direct responsibility for determination of whether or not an entity or the property of an entity is in compliance with Federal, State or local statutes, ordinances, laws, rules, or regulations, or for determinations or recommendations as to whether any license, authority to conduct business, or other certificate of authority issued by the City should be issued to an entity, be revoked, be canceled, or be suspended; or
 - 2. Direct responsibility for the approval of purchases by the City from an entity.
 - a. The term "entity" for the purposes of this Section means any individual, partnership, association, corporation (profit or non-profit), utility, or other organization, whether public or private, doing business with or subject to regulation by the City.
- B. Each employee exercising regulatory responsibility shall, within five working days of making application for employment with a person or entities doing business with or subject to the regulation by the City, report to his Department Director, in writing, that such application has been made. The term "application" means a verbal or written communication whereby an employee seeks employment or for a contractual relationship for remuneration.
- C. Each employee exercising regulatory responsibility that receives an offer of employment or an offer for a contractual relationship for remuneration from any person or entity doing business with or subject to the regulation by the City shall report such offer, in writing, to his Department Director within five working days of receipt of such offer. The term "offer" means a verbal or written communication in which a person or entity states an interest in retaining or employing the employee.
- D. Any employee exercising regulatory responsibility who has or who obtains a financial interest in an entity doing business with or subject to the regulation by the City shall report such fact to the appropriate Department Director within five working days of the acquisition of such financial interest. The term "financial interest", for the purpose of this Section,

means any arrangement whereby an employee acquires an ownership or material interest or the right to acquire an ownership of material interest, in an entity.

3.12 ADDITIONAL ETHICAL CONDUCT REQUIREMENTS FOR REPORTING INDIVIDUALS AND PROCUREMENT EMPLOYEES

A. As used in this Section:

- 1. The term "reporting individual" means any employee who is required by law, pursuant to Section 112.3145, Florida Statutes, to file full or limited public disclosure of his financial interests (financial disclosure). A procurement employee shall file financial disclosure pursuant to Section 112.3145, Florida Statutes.
- 2. The term "procurement employee" means any employee who participates through decision, approval, disapproval, recommendation, or preparation of any part of a purchase request; by influencing the content of any specification or procurement standard; by the rendering of advice; by investigation; by accepting quotes, by auditing or by participating in any other advisory capacity in the procurement of contractual services or commodities.
- B. A reporting individual or procurement employee shall report the receipt of expenses relating to any honorarium event from a political committee or committee of continuous existence, as defined in Section 106.011, Florida Statutes, from a lobbyist who lobbies the City directly or indirectly, or from the partner, firm, employer, or principal of such lobbyist.
- C. City employees shall not take part in any radio, TV, or video commercial for a political candidate where it is stipulated that the employee is a St. Pete Beach City employee, or the employee can be identified as a St. Pete Beach employee.
- D. This section shall not apply to members of the City Commission, any other elected City official, or to a member of any City board, commission or committee.

3.13 WHISTLEBLOWER POLICY

- A. Employees are expected to report any violation of law by any City employee, official, or business entity with which the City is doing business if such violation creates a serious and specific danger to the public's health, safety, or welfare.
- B. Employees are expected to report improper use of public office, waste of funds, or any other abuse or neglect of duty on the part of the City, a City

employee, or a member of the City Council or any City board. An employee with knowledge of such actions or activities shall report this information to his immediate supervisor, other members of supervision, Department Director, HR Administrator, or the City Manager for investigation and resolution. The investigation will be conducted by the City Manager, HR Administrator, and/or Department Director, depending on the nature and circumstances of the allegations. Information related to an ongoing investigation will be maintained on a confidential basis to the extent provided by Florida Statutes, Chapter 119.

C. Retaliation against an employee who reports any violation, abuse, or other improper action is strictly prohibited by Florida Statutes, Section 112.3187, and by City policy. Disciplinary action may be taken against an employee who knowingly files a false and/or malicious report.

3.14 USE OF CITY PROPERTY

A. General:

- The equipment and facilities that are provided for use by City employees are to be used to conduct official City business only. Employees shall not request or permit the use of City-owned vehicles, equipment, material, or property for personal convenience except when such services are approved by the City or are available to the public generally. Conducting personal business with City equipment/facilities or removing any property/equipment off-site for personal use is prohibited.
- 2. The operation, use, or occupancy as a passenger, of a City vehicle by any City employee while that City employee is consuming alcoholic beverages, is under the influence of alcoholic beverages, to the extent that his or her normal faculties are impaired, or is using any cannabis or controlled substance as defined in Chapter 893, Florida Statutes, or is under the influence of any such cannabis, controlled substance or narcotic to the extent that his or her normal faculties are impaired, is strictly forbidden.
- 3. Any City employee who has knowledge of the violation or use of City property by any other City employee should report such violation promptly to the City Manager, or to such employee's immediate supervisor.
- 4. Employees have no personal privacy rights when using City equipment, facilities, lockers, etc. In order to discourage the use and/or distribution of illegal drugs at any time or alcoholic beverages in the workplace, upon reasonable suspicion, searches for alcohol, illegal drugs or paraphernalia may be conducted on City

property or work sites, of employee's personal property stored of City property, including but not limited to any box, bag, or other container.

B. Communications Equipment:

- The City encourages the full use of the telecommunications systems to provide the best possible service to employees and our citizens, in accordance with the correct and permissible use of the system, as outlined below.
- 2. Office equipment used as a basis for telecommunications is described as, but not limited to, telephone, cell phones, voice mail, computer and beepers/pagers.
- 3. The use of all City telecommunications equipment is limited to legitimate City business purposes. Personal use of telecommunication equipment should be kept to a minimum. Under no circumstances shall City telecommunications equipment be used for any commercial promotional purpose, or to communicate any material of a political, religious, obscene, or derogatory nature.
- 4. Voice, text messaging and electronic mail are provided for City business communications only. However, incidental and occasional personal use may be permitted. Abuse of this privilege will be subject to disciplinary action.
- 5. Management and systems administrators may, from time to time, have a requirement to review employee voice, electronic or text messages for any of the following purposes:
 - a. To retrieve lost messages.
 - b. To recover from system failures or monitor system performance.
 - c. To comply with investigations into suspected criminal acts.
 - d. To ensure that City systems are being used for business purposes only, and in conjunction with City policy.
 - e. To fulfill public records requests.
- 6. Examples of Inappropriate Use of City Property (the following are prohibited):

- a. Use of the City mail system to receive personal mail or use of the City's postage meter for personal mail.
- b. Use of official letterhead stationery for personal correspondence.
- c. Use of obscene or profane language on City cellular telephones, mobile, or desk unit radios. "Borrowing" or taking City equipment from City premises for personal use.
- d. Use of City computers, electronic mail, or on-line services for personal business and use of software that is not owned or licensed by the City.
- e. Use of City equipment involving pornography and sexually oriented material.
- f. Inappropriate or excessive use of equipment where focus was oriented towards personal interest other than the interests of the City.
- g. Excessive time on the internet on sites or using software or applications that have nothing to do with the employee's job.
- h. The City reserves the right to monitor the use of any City property, equipment, facilities, lockers, etc., to ensure compliance with this policy.
- 7. There should be no expectation on the part of the employee of privacy with City equipment including but not limited to, telephone, cellular phones, electronic devices (i.e., tablets, etc.) computer equipment, beepers/pagers, etc. An employee who uses City property, equipment, or facilities for personal reasons or personal gain shall be subject to disciplinary action up to and including termination.

C. Use of City Vehicles:

- 1. Employees must possess and carry a valid Florida Operator's, Commercial Driver's License (CDL), Non-Commercial Driver's License, or Class "D" (Chauffeur's) license when operating a City vehicle or self-propelled machinery during work assignments as required by the position description.
- 2. Drivers and/or passengers of City vehicles must wear seat belts while the vehicle is in motion, in accordance with State law.

- 3. If an employee is involved in an accident while operating a City vehicle, he must notify his supervisor and contact a law enforcement agency immediately. If an employee receives a traffic citation as a result of the accident, payment of the ticket/fine shall be the employee's responsibility.
- 4. If an employee whose position requires a driver's license has his license suspended or revoked for any reason, it is the employee's responsibility to notify his supervisor as soon as he is aware of the suspension/revocation. The supervisor will review the circumstances involved. The employee may be placed on suspension, with or without pay, pending a review of the circumstances, or arrangements may be made to place the employee in a non-driving capacity depending upon circumstances of the license suspension/revocation, employee's overall record, and operational requirements. A memorandum outlining the circumstances will be forwarded to the HR Administrator for concurrence prior to any decision being reached/communicated to the employee.
- 5. The HR Administrator may monitor monthly, quarterly, or annual updates from the State of Florida Department of Highway Safety and Motor Vehicles regarding status of the driver's license of each employee whose job description requires possession of a valid Florida Driver's License. If such reports indicate that an employee's license has been suspended/revoked and the employee has failed to notify his supervisor as outlined above and has operated a City vehicle without a valid driver's license, in violation of this policy and in violation of law, the employee will be subject to disciplinary action, including dismissal.
- 6. Individuals who are not employees, vendors, contractors, etc., of the City are allowed as passengers in City vehicles **only** upon prior approval of the Department Director.
- 7. City vehicles are to be used to conduct official City business. Personal errands and/or business are not to be conducted while on City time or while using a City vehicle. However, employees who are authorized a take-home vehicle or a vehicle during work hours are permitted the use of the vehicle for local transportation during designated meal periods or work breaks while on duty and also when in route to and from work. No personal use is allowed after hours or on non-workdays unless for a City-sanctioned event.

- 8. Employees are not permitted to bring weapons into any City vehicle at any time (even if an employee has a valid concealed weapons permit), excluding sworn police officers who carry weapons in connection with their official duties. Employees are prohibited from carrying or having in their possession a weapon at any time while performing services on behalf of the City whether on or off City property, excluding sworn police officers who carry weapons in connection with their official duties.
- 9. Authorization for an employee to be assigned a take-home vehicle is in accordance with departmental policies and procedures, based on operational considerations, subject to approval of the Department Director.
- 10. Internal Revenue Code regulations govern the taxable or non-taxable status of take-home vehicles.
- 11. Failure to comply with this policy will result in disciplinary action, up to and including termination.

D. Employee Responsibility:

- Each employee who is provided City equipment, tools, supplies, or a vehicle shall exercise reasonable care in the safekeeping, use, and preservation of such City property and shall return the property in the same condition as the property was in when issued, except for normal wear and tear, upon separation from employment or upon request.
- 2. Employees shall promptly report, in writing, to their supervisor the loss, damage, or unserviceable condition of City property assigned to him or under his control. The immediate supervisor shall forward the report to the Department Director for appropriate action.

3.15 UNIFORMS, DRESS AND APPEARANCE

- A. Employees supplied uniforms by the City, are expected to wear uniforms in the performance of their job, shall report in a clean full uniform on each day worked. Uniforms must also be worn in the manner prescribed by the Department Director. Failure to comply may result in the employee being suspended for the day without pay. Repetition of such conduct shall subject the employee to further discipline.
- B. Employees are expected to observe normal and reasonable standards of personal hygiene and to present a professional appearance at all times. Failure to do so may result in the employee being sent home to correct the

- situation or for the day without pay. Repetition of such conduct shall subject the employee to further discipline.
- C. All hair, beards and mustaches must be of a length so as not to create operational or possible safety problems and must be maintained in a clean, neat and orderly fashion.
- D. Uniforms supplied by the City will be replaced by the City when they become unusable through normal wear and tear. A request to replace a uniform shall be submitted to the Director along with the damaged uniform.
- E. The employee is responsible to reimburse the City for uniforms lost or damaged through the employee's negligence and to return same upon cessation of employment.
- F. The City will withhold from the employee's pay reimbursement under E above up to the maximum allowed by applicable law.
- G. City issued uniforms may not be worn at times other than during the performance of City duties and during the normal trip to and from the employee's place of residency. No drinking alcohol in uniform before or after work hours.
- H. The employee shall be responsible for all laundering and minor repairs.
- I. Safety shoes, mandated by the employer to be worn on duty, shall be furnished without cost to the employee once a year, unless the Department Director determines in a particular case an employee's work responsibilities require an additional pair or pairs. Upon approval of the Employer, an employee shall be reimbursed for the purchase of safety shoes which meet or exceed the minimum quality standards provided the reimbursement does not exceed either the actual cost of the safety shoes or the actual cost of the standard, whichever is less. The Employer shall establish and post the standard for view by all employees at the start of each year. The City Manager will determine the maximum amount subject to budget availability and will review annually.
- J. Uniforms provided by the City
 - 1. Community Development: Field Staff short sleeve shirts, polo shirts, pants/shorts, and jacket
 - 2. Parking: Short sleeve/Long sleeve shirts, polo shirts, pants/shorts, and jacket.
 - 3. Pool: T-shirts, bathing suits, and sweatshirt.

- 4. Public Works: Field Staff -Short sleeve shirts, T-shirts (long/short), pants, shorts, steel-toes shoes and jacket.
- 5. Parks & Recreation
 - a. Recreation: T-Shirts, polo shirts, and jacket.
 - b. Parks: Short sleeved shirts, T-shirts (long/short), pants, shorts, steel-toes shoes and jacket.

3.16 GENERAL PROHIBITIONS

- A. Employees are expected to be aware that they are public service employees and to conduct themselves in a manner, which will in no way discredit the City, public officials, fellow employees or themselves.
- B. Employees shall avoid conduct or speech that is inconsistent with good order and discipline. They shall treat each other with the utmost courtesy and respect, and at all times refrain from making any derogatory remarks concerning each other. They shall direct and coordinate their efforts toward establishing and maintaining the highest level of efficiency, morale and achievement, and shall conduct themselves in such a manner as to bring about harmony among the various units of the City.
- C. No employee whose duties involve the use of a badge, card or clothing insignia as evidence of authority or for identification shall permit such badges, cards or insignia to be used or worn by anyone who is not authorized to use or wear them. Such badges, cards and insignia shall be used only in the performance of the official duties of the position to which they are related.

SECTION 4*

TYPES AND TERMS OF EMPLOYMENT

4.01 BASIS OF EMPLOYMENT

Employees are employed by the City as either probationary full-time, regular full-time, regular part-time, or temporary employees. (See Section 2).

4.02 PARTICIPATION IN BENEFITS

All regular full-time employees are eligible to participate in all City benefits. Employees working 30 hours a week or more are eligible for health insurance only. Other employees do not participate in employee benefit plans unless a particular plan so provides or unless required by law.

4.03 EMPLOYMENT TERMS

- A. Regular full-time employees who are not Directors or at-will shall continue as employees unless they are terminated, laid off, retire, or otherwise leave City employment as provided in these Rules.
- B. Full-time employees who have not successfully completed their initial probationary period as defined in Section 2 and all other employees of the City, including those covered by Subsection 1.02(B), serve at the will and pleasure of the City and may be disciplined or dismissed for any reason or no reason, subject only to applicable law. Such employees shall not have access to the grievance procedure set forth in these Rules or any applicable collective bargaining agreement unless the agreement provides otherwise. All decisions concerning their wages, hours, and working conditions shall be made by the City Manager.

4.04 REGULAR FULL-TIME AT-WILL EMPLOYEES - SPECIAL CONDITIONS

- A. Because regular full-time employees covered by Section 1.02(B) serve at the will and pleasure of the City Manager and have no property right in employment or their position, they shall be entitled to receive severance pay excluding statutory deductions if terminated by the City Manager.
- B. To the extent this provision does not violate then existing current law, Employees who have served in a position covered by Section 1.02(B) for more than three hundred sixty-five (365) consecutive days whose employment with the City is terminated, except for at will, subject to Subsections 4.04(C) and (D), be paid severance pay of one (1) week's pay for each year of continuous employment with the City but not less than eight (8) weeks nor more than twenty (20) weeks.

- C. To receive severance, the employee must execute a release satisfactory to the City.
- D. Severance under this subsection will not be paid to an employee terminated because of misconduct as defined in F.S. 443.036(29) (or as misconduct is defined in the law applicable at the time of termination), or because of dishonesty in connection with City business, for violation of the City drug-free workplace and alcohol policy, sexual and other illegal or improper misconduct, violation of the harassment policy, failure to fully or truthfully cooperate in an investigation conducted by or at the direction of the City, or failure to be acquitted if charged with a felony connected with City business.

4.05 PROBATIONARY EMPLOYEES

- A. Subject to 4.03(b), all regular full-time employees shall serve an initial probationary period of 2080 hours or 2912 hours for twenty-hour shift employees. Said probationary period may be extended by their Director for up to an additional ninety (90) days with the approval of the City Manager.
- B. When an employee is initially hired in a position which has a formal training program or requires certification or licensing, the probationary period as set forth in paragraph A above shall continue until the program, certification, or licensing is successfully completed.

4.06 PROMOTIONAL PROBATION

- A. An employee promoted to a higher level job classification shall serve an initial probationary period of 2080 hours or 2912 hours for twenty-four hour shift employees except where a license or certification is a prerequisite to holding the job, in which event the probationary period will continue until the license or certification is successfully obtained. During the promotional probationary period, the employee shall serve in the position to which he was promoted at the will and pleasure of the Director.
- B. In the event a promotional appointee is found to be unable or unwilling to perform the duties of the position to which the employee was promoted, the employee shall be returned to the position and status held immediately before the promotion, if the position is vacant; or the employee may be transferred to a vacancy within the same pay range subject to the approval of the City Manager. If there are no vacancies, the employee will be dismissed but may be considered for any future vacancies for which the employee is qualified.
- C. Extension of probation. Employees out on leave, while on probation will

have the probationary period extended, on a day-by-day basis equal to the amount of leave time taken. Employees shall not be removed from initial or promotional probation without the express, written approval by the Department Director and City Manager.

D. Accruals. Employees are entitled to accrue PTO during their probationary period. During probation employees may use PTO as it accrues.

4.07 LOYALTY OATH

As a recipient of public funds, and to the extent allowed by law, employees are required to sign a loyalty oath(s) as a condition of employment.

SECTION 5*

HIRING PROCEDURES

5.01 VACANCIES

- A. When a position vacancy occurs, the department submits a request to fill the position to the HR Administrator to initiate recruitment of a replacement. Requests for personnel should be submitted as soon as official notice of impending openings is received or need for a position is known.
- B. With the input of the Department Director, the HR Administrator prepares recruiting notices to publicize vacancies/solicit candidate applications.
- C. All positions will be posted internally for City employees for a minimum of ten days. Positions may also be advertised/posted externally concurrent with internal posting. Postings are "open until filled" or can be made for a specific time with a closing date.
- D. All persons inquiring about employment shall be directed to the HR page of the City website to complete and submit the online application form. Applications will only be accepted for posted vacant positions.

5.02 INTERVIEWS

- A. The HR Administrator will refer to the hiring Director qualified applicants for interview and assist in scheduling interviews, interviewing qualified candidates, and selecting the candidate who best matches the position requirements. In making this determination, the manager/supervisor shall consider each candidate's merit and ability, as evidenced by training, education, experience, certification, skills, etc., based on valid minimum standards/requirements outlined in the position description. Additionally, Veterans' Preference requirements shall apply in this process.
- B. Following completion of interviews, the hiring department completes the recommendation to hire form regarding candidate selection and returns the form and all applications to the HR Administrator for further action.

5.03 BASIS FOR SELECTION

- A. Employment with the City shall be based on skills, experience, training, education, ability, physical and mental ability to do the available work and other factors that are related to the performance of the job in question.
- B. As part of the pre-employment procedure former supervisors, employers and references provided by candidates shall be checked as a precaution against obtaining undesirable employees. Reference checks will be documented and made part of the applicant's file. Criminal background, driver license(s), if applicable, and social media accounts shall be checked also.
- C. The City reserves the right to reject any applicant for any reason or no reason subject only to applicable law.

5.04 TESTING

At its option and expense, the City may use legal physical, written or oral examinations and performance tests to assist it in the selection process.

5.05 DRUG TESTING

Subject to applicable law, the City may require submission to and successful passing of testing for the use of illegal controlled substances as a condition of employment or continued employment with the City.

5.06 DISABILITY AND MEDICAL EXAMINATIONS

- A. At the option of the City, applicants may be required to take a medical examination after they have been offered employment.
- B. If with the prior approval of the City Manager, an applicant is placed on the payroll prior to having completed a required medical examination, he shall be advised at the time he is placed on the payroll that his employment will be conditioned upon successfully passing a medical examination.
- C. Employees may be required to take a medical examination at any time by the City for reasons connected with their job (e.g., an accident on the job, fitness for duty).
- D. Employees who are directed to take a medical examination under paragraph A, B or C above and who refuse to do so will be automatically terminated.
- E. Applicants who take a medical examination pursuant to paragraph A, B or C above shall not be employed, or, if previously employed, shall be terminated immediately if the results of the medical examination show that

they are either mentally or physically unable to perform the essential functions of the job. However, if they have a legally recognized disability, they will be terminated only if they cannot be reasonably accommodated to perform the essential job functions of the job without undue hardship to the City and such action shall be subject to applicable federal, state and local laws dealing with handicap status.

F. Subject to applicable law, all medical examinations required to be taken under paragraphs A and B above shall include drug testing to determine the presence or absence of illegal controlled substance in their body. Drug testing above will be conducted under the Drug-Free Workplace and Alcohol Policy of the City.

5.07 PAY RATES FOR NEW EMPLOYEES

In an effort to successfully recruit the most qualified candidates for employment, the City of St. Pete Beach may recognize past years of comparable service in determining starting pay.

Compensation for at-will employees shall be determined by the City Manager at his discretion.

Wages are set forth in collective bargaining agreements except for MAPS employees. The pay grades for MAPS classifications are reviewed and subject to modification by the City Manager.

SECTION 6*

TYPES OF SEPARATIONS

6.01 TYPES OF SEPARATIONS

Separations and/or terminations from positions in the City service are designated as one of the following types. Personnel forms shall show the reason for the separation, and the last day worked. The effective date of the separation shall be the last day on which the employee is present for duty.

- A. Death Separation shall be effective as of the date of death. All compensation and benefits due to the employee as of the effective date of separation shall be paid to the designated beneficiary listed on the form for his city life insurance.
- B. Disability When an employee is determined to have a legally recognized disability which impairs his ability to perform the essential functions of the job, which he still cannot perform with reasonable accommodation and without undue hardship to the City, the City Manager may take whatever action he deems in the best interests of the City, including termination of employment.
- C. Dismissal or Discharge
 - 1. Temporary, part-time, and probationary employees are subject to dismissal from City employment pursuant to Section 4.03(B).
 - 2. Except as provided in Section 1.02(B), (C) and (D), regular full-time employees are subject to dismissal from City employment pursuant to Section 11.
- D. Reduction in Force (Layoff) Reductions in force shall be in accordance with Section 9.
- E. Probationary Termination When an employee is terminated for not successfully completing the probationary period.
- F. Resignation An employee wishing to leave the City in good standing shall file with his Director a written resignation, stating the date and reasons for his resignation. Such notice must be given at least two (2) weeks prior to the date of separation and the employee must work each regularly scheduled day during said two (2) week period to maintain good standing status. Employees who give such notice may be considered for reemployment with specific approval of the City Manager. A Director, with City Manager approval, may exempt an employee who has given less than the required notice if, in the Director's judgment, exceptional

circumstances warrant such an exemption. If proper notice is not given to the Director at least two (2) weeks prior to the date of separation employee is subject to forfeiture of any accrued sick leave or PTO balances.

G. Retirement - When an employee retires under the terms and conditions set forth for the City-sponsored applicable Pension/Retirement Plan.

6.02 RETURN OF PROPERTY AND FINANCIAL OBLIGATIONS

- A. At the time of separation from employment, the employee shall return all records, books, assets, uniforms, keys, tools and other items of City property to his department. Failure to return same in usable condition shall result in the maximum deduction allowed by law from the employee's final paycheck. Any balance due over and above the amount deducted from the employee's paycheck may be collected by the City through appropriate legal action.
- B. All outstanding voluntary debts to the City incurred by the employee, such as the cost of non-compensatory training, shortages or advance of leave or expense accounts, advances on pay and other standing debts due to the City will be deducted from the employee's final paycheck.
- C. All deductions under paragraphs A and B above shall be subject to the applicable State and Federal law.

TRANSFERS AND WORK OUT OF CLASSIFICATION

7.01 TEMPORARY TRANSFERS/WORK OUT OF CLASSIFICATION

- A. An employee assigned for thirty (30) calendar days or more to a position in a higher pay grade shall have a rate increase retroactive to the first day of the transfer. A 5% increase or the entry level of the temporary pay grade shall be given, whichever is greater. When the employee is reassigned to his original position, he shall be at the rate he would have achieved if no transfer had occurred.
- B. When a non-exempt employee temporarily works in a lower paid job classification, he shall receive the rate of pay for his regular job classification.
- C. A routine assumption of duties that occurs in the absence of another employee on vacation or with a short-term illness is not a transfer and does not affect salary.

7.02 PERMANENT TRANSFERS

An employee may be permanently transferred from one job classification or department to another job classification or department:

- A. At the employee's request, if in the opinion of the City Manager, it is in the City's best interest. An employee requesting a transfer shall submit a written request to the Department Director. Consideration for such action will depend on vacancy availability and other conditions which are deemed necessary as determined by the Department Director, and as approved by the City Manager.
- B. By the City Manager for operational or efficiency reasons.
- C. In all cases involving more than one (1) department, both Directors must agree to the transfer, unless in a particular case, the City Manager decides otherwise.

7.03 RECLASSIFICATION

When a job is reclassified, the employee holding the reclassified position will be paid in accordance with the applicable City Classification Plan and Section 8.08 at the grade assigned for the reclassified job.

A. Purpose – The most common type is the reclassification of an entire class of jobs. Reclassification can also take place when the work performed on

- a particular job changes substantially over a period of time, due to new technology or a change in the department's focus, by design or evolution.
- B. Request for Reclassification Reclassification can be initiated through independent, outside review conducted by or at the direction of the City or at the request of the individual or the supervisor, if accompanied by written supporting documentation. This documentation should be sufficient to support a reclassification, e.g., actual job duties and tasks.
- C. Approval Implementation of a requested reclassification of an individual or individuals to another job title within the existing Classification and Compensation Plan requires authorization of the City Manager. If the reclassification is for an entire class of jobs, it requires an amendment to the pay approved by the City Commission.
- D. Effect on Pay When a reclassification occurs, the employee is placed in the new grade.

PROMOTIONS/DEMOTIONS/LATERALS

8.01 NOTICE OF JOB VACANCY (OR POSTING)

Except at-will employees and when determined operationally necessary and efficient by the City Manager, all vacancies within the City will be posted for a minimum of ten (10) working days. Posted vacancies may also be advertised outside the City when deemed necessary by the City Manager. The posting will advise whether the job will be advertised or be initially restricted to employee applicants. The HR Office maintains the procedure used for posting positions.

8.02 APPLICATION

Employees who wish to be considered for the vacancy must complete a new online application found on the HR page of the City's website

8.03 POOL OF QUALIFIED APPLICANTS

The Director of the Department in which the vacancy exists, in conjunction with HR, will determine which of the employees, if any, who bid the job and outside applicants, when applicable, meet the minimum qualifications for the job.

8.04 INTERVIEW

All employee applicants determined by the Director and HR Administrator to meet the minimum qualifications for the job will be interviewed. When applicable, the most qualified from among outside applicants, if any, whom the City determines is better qualified than employee applicants will also be interviewed.

8.05 NO SUFFICIENTLY QUALIFIED APPLICANTS

If, after completing the interview and evaluation, the Director determines that none of the applicants are sufficiently well qualified for the job, the City Manager may fill the position in any manner he wishes.

8.06 BASIS OF SELECTION

- A. When the posting is restricted to City employees, in determining whom to promote from among qualified employee applicants, if any, the Director shall consider:
 - 1. The skills, knowledge and abilities to perform the job.
 - 2. The employee's past work-related experience with the City and elsewhere.

3. The employee's past performance record with the City.

When factors 1, 2 and 3 are relatively equal in the opinion of the City, time of continuous service in the Department and with the City, in that order, will be given preference.

B. When the posting is not restricted to City employees, the Director will consider factors 1, 2 and 3 for employees and factor 1, plus the outside applicant's references and past work-related experience and performance with other employers. When, in the opinion of the Director, all factors are considered relatively equal among all qualified applicants, City employees will be given preference, and as among them, time of continuous service with the Department and the City, in that order, will be given preference.

8.07 WAGE RATE

No employee shall be eligible for promotion to any higher grade until the employee has completed an initial probationary period, unless the City Manager approves the promotion.

8.08 PROMOTION, DEMOTION & LATERAL

- A. An employee who is promoted to a position in a higher pay grade, shall receive an increase in the rate of pay. The new pay shall be the minimum rate in the new pay grade or a 5% increase_whichever is higher.
- B. An employee who is demoted to a position in a lower pay grade shall receive a decrease in pay. The new pay shall be the top of the range in the lower pay grade or a 5% decrease in current salary whichever results in the lower rate of pay.
 - An employee may be demoted to a lower position for any of the following reasons: an employee's position is being abolished; an employee's position is being reclassified due to a lack of work; lack of funds; being removed during probation; an employee voluntarily requests a demotion; or for disciplinary reasons.
 - 2. A demoted employee will be required to serve a new probationary period and will be eligible for consideration for a merit pay increase on October 1st after the completion of the new probationary period, provided the employee is not at the top of the pay grade. All demotions must be at the direction of the Department Director concerned, reviewed by the HR Administrator for compliance with all related PRR and approved by the City Manager.
- C Lateral If an employee is moved to a position in the same pay grade, the rate of pay will remain the same and the anniversary date will not change.

SENIORITY LAYOFF AND RECALL

9.01 ACCRUAL

City, departmental and job classification seniority shall continue to accrue during all types of compensable leave approved by the City. Approved leaves of absences of thirty (30) or more consecutive workdays without pay shall not count towards the accrual of classification seniority unless the law requires otherwise.

9.02 LOSS OF SENIORITY

An employee shall lose his seniority and be terminated from employment as the result of any one of the following:

- A. Discharge.
- B. Retirement.
- C. Voluntary resignation.
- D. Layoff exceeding one (1) year.
- E. Failure to report to the Director the intention to return to work within three (3) business days of receipt of a recall offer notice.
- F. Failure to return from military leave within the time limits prescribed by law or any other leave unless an extension has been approved in advance by management.

9.03 LAYOFF SELECTION

In the event the City decides to lay off employees within a department, the City will first lay off those employees employed on a part-time, temporary or probationary basis. If further layoffs are necessary, selection among regular full-time employees shall be based upon:

- A. Ability to perform all skills, knowledge and abilities of the work available.
- B. Special skills essential to the performance of the available work.
- C. Job performance as reflected by the performance evaluations for the past three (3) years or the most recent evaluations available.
- D. Departmental classification seniority.

E. Classifications performing work that is no longer needed.

When, in the opinion of the Director, factors A, B and C are relatively equal among employees, factor D shall prevail.

The City Manager reserves the right to determine the knowledge, skills and abilities of the positions impacted

9.04 PERMANENT LAYOFFS

In some cases, the City may utilize a layoff under circumstances where there is no reasonable expectation to return to work. Such layoffs will be designated permanent and the employees laid off shall not be eligible for recall.

- A. Full-time employees who have completed their initial probationary period and who are scheduled to be laid off for lack of work, funds or other reasons where there is no fault on the part of the employee shall be eligible to receive severance pay as follows:
 - One (1) week of pay at their straight time hourly rate or salary, whichever applies, less statutory deductions, for each full year of service as an employee of St. Pete Beach, capped at twelve (12) weeks.
 - 2. The employee's last annual performance evaluation must be satisfactory or better and the employee must be on active duty not on leave of absence or suspension without pay.
 - 3. The employee must execute a release of all claims, including the right to file a grievance under any applicable CBA, the City PRR, as well as any and all judicial and/or administrative claims. If an employee is in a collective bargaining unit, the collective bargaining agent must approve the employee's release of his/her right to file a grievance under the applicable collective bargaining agreement and the City PRR.
 - 4. Employees covered by Section 1.02(B) are not entitled to severance pay under this Section, but as provided in Section 4.04.
- B. Eligible employees not covered by Section 1.02(B) or (C) who have recall rights under this PRR or any CBA, may elect to retain recall rights in lieu of severance pay as provided in Section 9.04(A) above.

9.05 RECALL

Except for employees laid off pursuant to Section 9.04 above regular full-time employees who are recalled by the City within twelve (12) months shall have their City service, departmental service, and job classification seniority restored:

however, they will not be given credit for the period of the layoff nor shall they receive wages or benefits during the period of the layoff.

9.06 DECISIONS FINAL

Decisions made pursuant to this Section shall be final and shall not be subject to the grievance procedure in Sections 12 or 13.

SECTION 10*

ATTENDANCE/TARDINESS

10.01 PRESENT AND ON TIME

All employees are expected to report for duty at the scheduled time and remain there until the scheduled leaving time. Each Director shall be responsible for the attendance and timeliness of all persons within his department.

10.02 CALL-IN

Employees are required to call or text before they are scheduled to report to work when they are going to be absent or late. (Check with your Supervisor or Director for specific instructions that pertain to your department). Hourly employees who fail to call in before the employee's shift begins will be subject to discipline, unless the Director is satisfied that the failure to call in was for a reason beyond the employee's control. Such instances will be recorded as unscheduled PTO.

10.03 VERIFICATION

The Director may require an employee to establish to his satisfaction that an absence or tardiness was for a legitimate reason. Such proof, in the case of sickness or injury, may include the presentation of a medical doctor's excuse from a doctor acceptable to the City.

10.04 CONTINUING ABSENCE

In the case of a continuing absence, the employee must call in each day unless otherwise instructed by his Director or Supervisor.

10.05 PERSON TO CALL

Call-ins are to be directed to the employee's immediate Supervisor; however, in the event the immediate Supervisor is not available, the employee must speak with the Director, or his designee. If no one is available to speak with the employee, the employee must leave a contact number where they can be reached.

DISCIPLINARY ACTION

11.01 GENERAL STATEMENT

It is the hope of the City that effective supervision and employee relations will avoid most difficulties which otherwise might necessitate discipline of employees. However, when disciplinary action becomes necessary, the City recognizes the fact that each situation differs in many respects from others that may be similar in some ways. Thus, the City retains the right to treat each incident on an individual basis without creating a precedence for other cases which may arise in the future as to a particular employee or group of employees and to determine the appropriate discipline in every matter on a case by case basis. Note that this Section 11 does not apply to employees who serve at the will and pleasure of the City Manager.

11.02 FORMS OF DISCIPLINE

- A. The City recognizes the following types of discipline:
 - 1. Verbal reprimands.
 - 2. Written reprimands.
 - Suspension without pay.
 - 4. Demotion.
 - 5. Combination of the above.
 - 6. Termination of employment.
- B. A Director may also combine a probationary period not to exceed six (6) months with all forms of discipline except termination.
- C. The Director, or his designee, will consult with HR in deciding appropriate disciplinary action greater than a written reprimand. However, the Director may place the employee on administrative leave with pay to the next regular workday of the City (Monday through Friday) if the Director deems it is operationally necessary. The Director shall then immediately consult with HR, to determine whether temporary suspension without pay or additional administrative leave is appropriate pending completion of an investigation and a final decision as to the appropriate disciplinary action, if any.

The City Manager will determine if at-will employees will continue administrative leave past the first day and if so, whether it will be with or without pay.

11.03 APPLICATION OF DISCIPLINARY ACTION

- A. Verbal and written warnings for regular full-time employees may be given for any reason listed in Section 11.05, Group I or Group II, or for any other just cause.
- B. Subject to Section 11.02(C), regular full-time employees may be suspended without pay temporarily pending investigation and final determination of possible disciplinary action.
- C. Subject to Sections 1.02(B), (C) and (D), 3.14(A), 4.05, 11.03(B) and 11.06(A), regular full-time employees who have completed their initial probationary period may be suspended without pay, demoted for disciplinary reasons or terminated for any reason listed in Section 11.05 below, or for any other just cause.
- D. Before an employee is temporarily suspended without pay if the employee is available, the Director, or his designee, will advise the employee of and the reason for the temporary suspension and give the employee the opportunity to explain his position verbally or in writing.
- E. All other employees serve at the will and pleasure of the City Manager and may be subject to discipline, up to and including termination, as he deems appropriate subject only to applicable law.

11.04 NOTICE OF DISCIPLINARY ACTION AND PRE-DISCIPLINARY PROCEDURE

- A. *Pre-Disciplinary Procedure*. When a Director is considering suspension without pay, other than temporary pending a final decision, demotion or termination of a regular full-time employee who has completed his initial probationary period:
 - 1. The employee shall be given written notice of the reason such disciplinary action is being considered, the names of the witnesses, a summary of the information on which the Director relies and an opportunity to present his position either verbally or in writing to the Director before the decision is made. Notice shall be complete upon either personally handing it to the employee or mailing it to the employee's address as contained in the City personnel file.
 - 2. The Director, with the approval of the City Manager, may suspend the employee without pay immediately pending the ultimate decision, provided the employee shall be given an opportunity to explain his position before the temporary suspension without pay.

3. If after considering all of the evidence before him, the Director decides to suspend the employee without pay, demote or terminate, he shall notify the employee in writing of the decision and the reasons therefore, and advise the employee of his right to a due process hearing under Section 12.02 – Appeal Procedure.

In all cases of written reprimand, probation, suspension without pay, demotion, any combination of same, or termination; the regular full-time employee will be notified in writing of the action taken and a copy of such notice shall be retained by the City in the employee's personnel file. If the employee is exonerated the City will place a memorandum in the file stating that the employee was exonerated, and the employee shall be provided back pay, less interim earnings, and deductions for periods the employee was unable to work due to the temporary suspension.

11.05 TYPES OF OFFENSES

There are two (2) groups of example offenses for which full-time employees may be disciplined up to and including termination, and the guidelines for recommended penalties for those examples of unacceptable conduct are set forth below; however, the principles concerning application of discipline to these sample offenses or others as set forth in Sections 11.01-11.02 above shall apply. Nothing herein shall be construed to limit disciplinary action to the sample offenses enumerated below, and suspension without pay, demotion or termination may be for any just cause.

This paragraph provides recommended but not mandatory penalties to apply to the specific example offenses listed here; however, the penalty utilized shall be discretionary with management in all matters of discipline and nothing herein shall require that a particular form of discipline be utilized in any case prior to the utilization of another form of discipline.

GROUP 1 OFFENSES

First Offense – Verbal or written reprimand Second Offense - Up to ten (10) days suspension without pay Third Offense - Up to and including termination

- 1. Wasting time, loitering or leaving assigned work area during working hours without permission.
- 2. Excessive tardiness and/or absenteeism, which disrupts departmental operation regardless of the reason.
- Taking more than allowable times for meal or rest periods.
- 4. Unacceptable, inefficient productivity or competency.

- 5. Sleeping on the job unless authorized to do so.
- Violating a safety rule or practice; not wearing required safety clothing or equipment.
- 7. Engaging in horseplay, scuffling, wrestling, throwing things, malicious mischief, distracting the work of others, catcalls, or other disorderly conduct.
- 8. Failure to report the loss or damage of City equipment or other City property entrusted in the employee's custody.
- 9. Failure to keep the City and department notified of the employee's current proper physical address and telephone number.
- 10. Gambling, lottery or engaging in any other game of chance in any fashion, while on duty or that brings disrepute upon the City.
- 11. Violation of published City or departmental policies, rules, standard, orders, operating procedures or regulations.
- 12. Unexcused tardiness or absence.
- 13. Violation of the Standards of Conduct in Section 3.

GROUP II OFFENSES

First Offense - Up to and including termination

- 1. Conviction of a felony.
- Abuse of leave privileges.
- 3. Use of official position for personal advantage.
- 4. Deliberately or negligently misusing, destroying, losing or damaging any City property or property of an employee.
- 5. Falsification of personnel, City, or Departmental records, including employment applications, accident records, work records, purchase orders, time records, or any other report, record, or document.
- 6. Making false claims or intentional misrepresentation in an attempt to obtain sickness or accident benefits, workers' compensation, or any other benefit.

- 7. Insubordination or the refusal to perform work assigned, or to comply with written or verbal instructions of a Supervisor.
- 8. Use or possession or display of firearms, explosives, or weapons on or in City property unless authorized.
- 9. Removal of City property or any other employee's property from City locations without proper authorization, theft of City property or any employee's property.
- 10. Failure to return at the end of an authorized leave of absence.
- 11. Concerted curtailment, restriction of production, or interference with work in or about the City's work stations including, but not limited to, instigating, leading, or participating in any walkout, strike, sit down, stand-in, slowdown, or refusal to return to work at the scheduled time for the scheduled shift.
- 12. Absent without permission or leave (AWOL)/Job abandonment.
- 13. Acceptance of a gift, service, or anything of value in the performance of duty or under any other circumstances where the employee knew or should have known it was given with an expectancy of obtaining a service or favored treatment.
- 14. Possession, use, sale, attempt to sell, or procure illegal controlled substances at any time whether on or off City property or whether on or off duty; and possession, use, sale or attempt to sell or procure alcoholic beverages while on duty, on City property, or while operating or riding in or on City equipment. Police officers transporting controlled substances or alcohol as evidence are excluded from this provision.
- 15. Refusal to fully and truthfully cooperate in an investigation conducted by or at the direction of the City.
- 16. On or off the job conduct which adversely affects the ability of the employee to perform his duties and/or adversely affects the efficient operation of the City government or any department, division, or area of City government.
- 17. Discourteous, insulting, abusive, or inflammatory language or conduct toward the public or co-workers.
- 18. Improper harassing comments, any other form of harassment or any improper act directed to any City employee or the general public.
- 19. Threatening, intimidating, coercing, or interfering with fellow employees or supervision at any time.

- 20. Provoking or instigating a fight or fighting while on duty.
- 21. Unauthorized personal use of the City's exempt tax number for any reason.
- 22. Accepting a bribe or gratuity, committing an illegal act or accepting a gratuity for performing the normal duties as a City employee.
- 23. Failure to report in writing an offer of a bribe or gratuity to permit an illegal act.
- 24. Communicating or imparting confidential information either in writing or verbally to any unauthorized person.
- 25. Refusal to sign an acknowledgment of receipt of disciplinary action.
- 26. Failure to possess and maintain a current and valid state motor vehicle operator's license, if driving a vehicle is required by the City as an essential part of the employee's job.
- 27. Failure to immediately report an arrest or conviction of a DUI or DWI to the City.
- 28. Loss of a license or certification required by the City, the State or other governmental entity to perform the job for which the employee is assigned.
- 29. Failure to return to light duty when assigned to do so.
- 30. Disclosure of medical information which violates HIPAA regulations.
- 31. Failure to immediately report an on-the-job accident or personal injury.
- 32. Reporting to work or working while unfit for duty, either mentally or physically, unless the condition is a legally recognized disability in which case the matter will be dealt with in accordance with applicable law.
- 33 Physical threats or violence.

The above list does not include all of the reasons for which an employee may be subject to disciplinary action, but as stated earlier, is intended to provide examples of inappropriate conduct.

11.06 SUSPENSION PENDING RESOLUTION OF CRIMINAL CHARGES

A. In the event an employee is charged with any crime, the employee may be suspended with or without pay.

- B. At any time, the City Manager shall have the option of taking disciplinary action based on his own investigation without regard to the existence, status or final disposition of the criminal charges.
- C. The City Manager may elect to wait until the criminal proceeding, or a particular phase thereof is concluded before considering disciplinary action. In such a case, the City Manager may take the resolution of the criminal proceeding, or phase thereof, under consideration but shall not be bound thereby and shall make its determination as to the facts and the appropriate disciplinary action, if any.
- D. Under paragraphs B and C above, the City Manager will not consider anything less than a finding by a judge or jury, whichever is applicable, of not guilty as relevant to the issue of whether the employee engaged in the conduct in question.
- E. If an employee charged with a crime is found not guilty by a judge or jury, and the City Manager determines no disciplinary action is warranted, the employee will be reinstated with back pay less amounts earned, unemployment compensation and periods of time the employee was unavailable to work or did not make every reasonable effort to find work.

APPEAL PROCEDURE FOR DISCIPLINARY ACTIONS/NAME CLEARING HEARINGS FOR TERMINATED EMPLOYEES

12.01 WRITTEN WARNINGS

Appeals of written warnings shall be under PRR Section 13.

12.02 SUSPENSIONS WITHOUT PAY, DEMOTION AND TERMINATION APPEAL PROCEDURE

- A. A regular full-time employee who has successfully completed his initial probationary period who is suspended without pay (other than temporarily), demoted for disciplinary reasons or terminated who wishes to appeal must appeal in writing to the City Manager within ten (10) working days after notice of his suspension without pay, disciplinary demotion or termination.
- B. Upon receipt of a timely appeal, the City Manager shall arrange a hearing at which the Director shall be required to establish just cause by a preponderance of the evidence.
- C. Not less than five (5) calendar days prior to the hearing, to the extent not already provided under Section 11.04 the Department Director and the employee will provide the other with a list of witnesses, except rebuttal witnesses, a brief summary of their anticipated testimony, and a list of all documents, except rebuttal documents, upon which they intend to rely.
- D, The employee shall be entitled to be represented by counsel of his choice at his expense and shall have the right to present evidence, examine and cross-examine witnesses and state his position orally or in writing.
- E. Proceedings shall be recorded by tape, video or other recorder or by court reporter.
- F. The City Manager shall consider the evidence before him and make Findings of Fact and Conclusions of Law, which shall be final and binding on all concerned.

12.03 GENERAL PROVISIONS

A. The time limits set forth above, may be extended upon written request for reasons considered appropriate by the City Manager. Failure of an employee to file an appeal in a timely fashion, unless an extension has been granted in advance, will constitute an automatic abandonment of his appeal.

B. In the event a grievance is filed which involves two (2) or more employees in the same or similar event, happening or condition, the City Manager may rule that all grieving parties shall be governed by one (1) grievance form and resulting decision. The Department Director, City Manager need not rule on each individual grievant's case separately when the foregoing conditions are present.

12.04 DELEGATION OF AUTHORITY

At his option, the City Manager may delegate to another Director, an attorney or other person to hold the hearing and make the final decision or hold the hearing and make recommended Findings of Fact and Conclusions of Law, in which event the City Manager shall be bound by the recommended findings of fact as long as they are supported by probative evidence in the record and conclusions of law. The City shall bear the fee of any substitute for the City Manager which he delegates.

12.05 NAME CLEARING HEARING*

- Any employee who is terminated and the former employee contends that, Α. in relation to the termination, false statements that are damaging to the employee's reputation have been made public or placed in the former employee's personnel file, the former employee is entitled to a Name Clearing Hearing. The purpose of the hearing is to afford the former employee an opportunity to clear his/her name by responding to the statements. The hearing will not serve as any type of appeal of the termination or to gain reinstatement. Every former employee is NOT entitled to a Name Clearing hearing. Such a hearing is only required when any statement relating to termination "stigmatizes" the former employee's reputation and the former employee denies the truth of the statements. A statement is stigmatizing if it involves allegations of dishonesty, immortality or other conduct which may damage the former employee's reputation among associates and impair his/her ability to obtain other employment. Examples include: falsifying records, misappropriating City property or funds, mental instability or statements that the employee is untruthful, untrustworthy, unethical or has demonstrated poor judgment, or has committed a crime or immoral act.
- B. A request for a Name Clearing Hearing must be made by the employee in writing within 14 calendar days from the effective date of termination. If the employee has appealed his/her termination as set forth above, the employee has 14 days after receiving the City Manager's Findings of Facts and Conclusions of law that uphold the termination to make a request for a Name Clearing Hearing. Failure of a former employee to request this hearing on a timely basis is a waiver. Any request for a Name Clearing Hearing must identify the specific statements the former

employee contends are false. At the hearing the former employee will be allowed to present testimony from him or herself and witnesses and documents to support the former employee's allegations that the statements at issue are untrue. Either the City Manager or a person designated by the City Manager will act as the hearing officer. The hearing officer will not be the former supervisor of the former employee.

C. The hearing will not consist of or result in the formulation of any conclusions or the nullification of the termination. The sole purpose is to provide an opportunity for a former employee to present information to clear his/her name. Within 30 days after the hearing, the hearing officer will issue a written report summarizing the evidence presented. A copy of the report will be placed in the former employee's personnel file along with a transcript of the hearing provided that the former employee has provided the court reporter for the hearing and provided a copy of the transcript to the City.

EMPLOYEE GRIEVANCE PROCEDURE – WRITTEN WARNINGS AND NON-DISCIPLINARY MATTERS

13.01 PURPOSE

It is the purpose of this grievance procedure to assure employees that written warnings and non-disciplinary problems and complaints arising under the PRR will be considered fairly, rapidly and without reprisal. It is expected that the procedures set forth below will encourage employees to discuss with their supervisor matters pertaining to conditions of employment as they affect individual employees. In addition, free discussion between employees and supervisors will lead to better understanding of practices, policies, and procedures which affect employees. This will serve to identify and eliminate conditions which may cause misunderstandings and grievances.

13.02 DEFINITION OF A GRIEVANCE

A grievance is a complaint about a written warning on the misapplication or misinterpretation of these rules or applicable Departmental rules and regulations. Discipline except written warnings shall not be considered under this Section, but only under Section 12.

13.03 PROCEDURE

- A. Step one: An employee shall present his complaint to his immediate Supervisor within five (5) working days from the time of occurrence of the problem. The Supervisor shall attempt to resolve the problem within five (5) working days after the complaint is made to him.
- B. Step two: If the employee has not received an answer from the immediate supervisor within five (5) working days, or if the employee feels the answer received is not satisfactory, he will put in writing the facts and circumstances of the problem and present the written statement to his Director within five (5) working days after the Supervisor's deadline in Step one. Assistance will be provided by HR, if requested, including for those employees who cannot read or write or have a language problem. The Director will investigate the grievance and meet with the employee to discuss the grievance within five (5) working days. The Director will notify the employee of his decision within five (5) working days following the meeting date.
- C. Step three: If the employee has not received an answer from the Director within five (5) working days, or if the employee feels the answer received is not satisfactory, he will put in writing the facts and circumstances of the problem and present the written statement to the City Manager within five

(5) working days after the Director's deadline in Step two. Assistance will be provided by HR, if requested, including for those employees who cannot read or write or have a language problem. The City Manager will investigate the grievance and meet with the employee to discuss the grievance within five (5) working days. The City Manager will notify the employee of his decision.

HOURS OF WORK AND OVERTIME

14.01 HOURS OF WORK*

- A. The City shall establish the hours of work in accordance with the operational needs of the City.
- B. The Directors shall schedule the work as necessary to provide full service but should attempt to avoid overtime work except where operationally necessary.
- C. Unless otherwise provided by a CBA, employees shall be scheduled for an unpaid lunch break of not less than one-half (½) hour. Employees requesting a longer lunch break must get approval from their Director and will not be paid for the time.

14.02 REGULAR WORKWEEK OR PERIOD

Except for employees on a Section 7(K) schedule in the Fire Department or as otherwise provided by a CBA:

- A. The regular Workweek for regular full-time employees shall be forty (40) hours in a seven (7) day period. The workweek is Midnight Sunday to 11:59 p.m. the following Saturday. The City Manager may establish the basic work schedule and hours of work best suited to meet the needs of the departments and the City to provide proper service to the community. Nothing in these rules shall be construed as a guarantee or limitation of the number of hours to be worked per week.
- B. The basic work schedule shall be from Monday through Friday of each week unless specified or scheduled by the City Manager to meet the particular requirements of the City or individual departments. When the City Manager deems it necessary, work schedules may be established other than the basic Monday through Friday schedule.
- C. Lunch and break periods are scheduled at the discretion of the Director or his designee.

14.03 OVERTIME*

Except for employees on a Section 7(K) schedule in the Fire Department:

A. Hourly employees shall be paid at a rate of one and one-half (1½) their regular hourly rate when they work more than forty (40) hours in the Workweek period unless they take comp time pursuant to (C) below.

- B. There shall not be any duplication of overtime or premium pay.
- C. Hourly employees may be allowed or required to take comp time in lieu of receiving overtime pay up to a maximum of forty (40) hours per year. At the end of the fiscal year, unused comp time will be paid to the employee. Unused comp time will not be carried over into the next fiscal year. An employee's comp time balance shall be paid in accordance with applicable law upon cessation of employment. An employee's request to use comp time on the date requested may be denied if it would unduly disrupt the operations of the City.
- D. When the governor of the State of Florida, Mayor of St. Pete Beach or representative of another local jurisdiction officially declares a state of emergency covering St. Pete Beach, employees performing essential services shall be required to work as deemed necessary by the City Manager.

Pay for time worked during emergency conditions will be as follows:

- 1. FLSA exempt employees shall be paid additional compensation above their regular salary for work during any week in which an emergency has been declared and they work in excess of 50 hours.
- 2. All non-exempt MAPS employees will be paid time and half for all hours worked in excess of their normal work week.

Employees on official paid or unpaid leave e.g. PTO, family or medical leave, etc. during an emergency, will not be eligible for any excused absence declared as a result of the emergency or emergency pay during the official leave.

14.04 ASSIGNMENT AND WORKING OVERTIME

- A. Overtime will be authorized or directed only when it is in the interest of the City and is the most practicable and economical way of meeting workloads or deadlines. Employees are to work only overtime as authorized but must report all hours worked to ensure compliance with the FLSA regardless of whether the work is authorized or not. Employees who work overtime without receiving authorization in advance may be disciplined.
- B. Employees are required to work overtime when assigned unless excused by their Supervisor. An employee desiring to be excused from overtime work assignments shall submit a request to the immediate Supervisor who shall rule on the request.

14.05 HOURS COUNTED

Hours of absence due to paid conference/training time, PTO, or jury duty will NOT be counted as hours worked for the purpose of determining eligibility for overtime. Only actual hours worked shall be counted for determining if overtime is worked.

14.06 CALL BACKS

When an hourly regular full-time employee who has worked his regular shift or more, is released to go home, leaves the City premises, and is called back to work, the employee shall be paid at the applicable hourly rate for three (3) hours, or the actual hours worked, whichever is greater.

14.07 TELECOMMUTING

Depending on the nature of the position, the City Manager at his discretion may authorize a temporary or longer-term telecommuting arrangement, subject to a formal policy.

HOLIDAYS

15.01 LIST OF LEGAL HOLIDAYS/DAYS OBSERVED

A. The City recognizes the following holidays:

New Year's Day
Martin Luther King Day

Veterans Day
Thanksgiving

Presidents' Day Friday after Thanksgiving

Memorial Day Christmas Day

Independence Day (Religious Holiday substitution)

Labor Day Personal Day

B. Eligibility for Holidays:

- 1. Regular Full-Time employees receive 8 hours of holiday pay per holiday.
- 2. Regular Part-Time employees who work an average of 20 hours per week or more will receive 4 hours of holiday pay per holiday.
- 3. Part-Time Temporary employees are not eligible for holiday pay.
- C. All MAPS employees shall be granted two (2) additional Personal Days per year which are to be used by September 30th of each year or will be lost. New Hires will not be eligible for the additional two (2) Personal Days until they have completed six (6) months employment. Effective October 1, 2020 the sixteen (16) hours for the two additional Personal Days will be discontinued and will be added to the Paid Time Off accruals.
- D. When a holiday falls on a Saturday or Sunday, the following Monday or the preceding Friday will be declared a holiday for City employees, as determined by the City Manager.
- E. A Personal Day must be taken a full day at a time and must be approved in advance by the Department Director, or his designee.
- F. The Personal Day(s) shall be treated as a holiday and therefore is not subject to accrual; and if not taken, will not be paid out at time of separation of employment with the City. All Personal Days shall be used within the fiscal year, unless Departmental management denies the request due to scheduling problems, in which event the employee shall be compensated for Personal Days to which he was entitled but could not use at the end of the fiscal year. Personal Days shall be requested

- through Time Clock Plus prior to the chosen day and are subject to approval by the Department Director.
- G. Employees shall also enjoy special holidays that are observed by the City during the term of this Agreement. Special holidays are defined as nonregularly scheduled holidays established by the City Commission to commemorate a special event or occasion not regularly provided to employees.

15.02 REQUIREMENTS TO RECEIVE HOLIDAY COMPENSATION

- A. Subject to 15.01 (B) above, all eligible employees will receive one (1) workday off with pay for each of the holidays.
- B. An employee must be on active pay status in order to qualify for the holiday time.
- C. Employees who are on PTO or other leave with pay during the holiday will receive holiday compensation. Employees who are required by their Supervisor to work on the day observed as a holiday must work that day to be eligible to receive holiday pay. An employee who is scheduled to work on the day observed as a holiday and reports sick will be charged with PTO for that day subject to Section 15.03.
- D. Employees scheduled to work on a holiday, and who in fact do work, shall receive pay at the normal straight rate or overtime, whichever applies, for the actual number of hours worked that day, plus a normal day's pay for the holiday provided they meet the eligibility requirements.
- E. Employees whose normal day off from work falls on a holiday shall receive holiday pay at the normal straight rate for the day, provided they meet the eligibility requirements.
- F. When a holiday falls on a day a part-time employee is not scheduled to work and the employee does not work, the employee shall receive four (4) hours pay at the straight time rate for the holiday.

15.03 ABSENCE DUE TO SICKNESS

An employee scheduled to work a holiday who fails to work because of sickness or injury shall not receive holiday pay unless (1.) he notifies his Director at least one (1) hour before he is scheduled to report for work and (2.) upon request, he presents evidence satisfactory to the Director, which may be a medical doctor's excuse, that his absence was due to a bona fide, unforeseen serious illness or injury. The employee who fails to follow this procedure will also be subject to disciplinary action up to and including termination. The Director may excuse the first requirement if he is convinced that failure to notify as required was for a reason clearly beyond the employee's control.

15.04 TWENTY-FOUR HOUR SHIFT EMPLOYEES

Regular Full Time Employees on 24 hours shifts shall receive a bank of 288 at the beginning of the fiscal year in lieu of having Holidays and Personal Days.

OR

- A. Regular Full Time Employees on a twenty-four (24) hour shift schedule will receive a twenty-four (24) hour shift off with pay every thirteenth (13) shift in lieu of holiday pay, but no more than nine (9) shifts in any fiscal year.
- B Regular Full Time Employees on a twenty-four (24) hour shift schedule will be entitled to seventy-two (72) hours for personal time. Personal time must be taken a full day at a time and must be approved in advance by the Department Director, or his designee. The personal time shall be treated as a holiday and therefore is not subject to accrual. All personal time shall be used within the fiscal year, unless Departmental management denies the request due to scheduling problems, in which event the employee shall be compensated for personal time to which he was entitled but could not use at the end of the fiscal year. Personal time shall be requested through Time Clock Plus prior to the chosen day and is subject to approval by the Department Director.

PAID TIME OFF (PTO) POLICY

16.01 DEFINITION

Paid Time Off (PTO) is an all-inclusive flexible time off policy in place of traditional individual vacation, sick, and injury leave programs. It does not apply to Jury Duty or bereavement leave. PTO is an employee benefit. It is a program to allow employees an established amount of paid absence without regard to the reason.

16.02 ELIGIBILITY AND USAGE

- A. All Regular Full-Time employees will be eligible to accrue PTO.
- B. All Regular Part Time employees will accrue PTO based on hours worked.
- C. Temporary employees are not eligible.
- D. PTO leave may be used for the following purposes (subject to approval in paragraph 16.05 below):
 - Vacation leave
 - 2. Sick leave
 - 3. FMLA leave
 - 4. Personal leave for any number of reasons, such as:
 - a. Medical and dental appointments and treatment which is necessary during working hours.
 - b. Absences for personal business which cannot be conducted during off duty hours.
 - c. Holidays other than those observed by the City as official holidays.
 - d. Leave of Absence Without Pay
 - e Military leave
 - 5. To supplement any leave for which employee is receiving compensation such as leave covered by short term disability benefits or workers compensation benefits only to the extent necessary to make up the difference in all compensation received from all sources at the employee's straight time weekly earnings or salary whichever applies.

16.03 ACCRUAL

- A. Full-time employees shall accrue PTO leave each payroll as follows:
- B. New hires will be eligible to begin accruing as of their date of hire. Hours available may be used with no waiting period.

Completed Continuous Months of Service	Bi-weekly Accrual	Annual Accrual	Max Accrual End of FY
0 to 59 Months	4.923 hours	128 hours/16 days	240 hours
60 to 119 Months	6.461 hours	168 hours/21 days	280 hours
120 + Months	8.000 hours	208 hours/26 days	320 hours

For MAPS Employees in lieu of Two additional Personal Days effective 10-1-2020

Completed Continuous Months of Service	Bi-weekly Accrual	Annual Accrual	Max Accrual End of FY
0 to 59 Months	5.539 hours	144 hours/18 days	240 hours
60 to 119 Months	7.077 hours	184 hours/23 days	280 hours
120 + Months	8.616 hours	224 hours/28 days	320 hours

C. Employees working twenty-four hour shifts shall accrue PTO as follows:

Completed Continuous Months of Service	Bi-weekly Accrual	Annual Accrual	Max Accrual End of FY
0 to 59 Months	7.385 hours	192 hours	360 hours
60 to 119 Months	9.692 hours	252 hours	420 hours
120 + Months	12.000 hours	312 hours	480 hours

16.04 APPROVAL

- A. Scheduled PTO In order to ensure effective operational scheduling PTO should be requested as far in advance as possible, but in no event less than one (1) work day before the leave is to commence unless the failure to make a timely request is determined by the employee's Department Director to be for reasons beyond the control of the employee.
- B. Unscheduled PTO In the case of unforeseen sickness or injury of the employee or an immediate family member, the employee must advise his/her supervisor or Department Director as soon as possible, but not later than one hour before the employee's scheduled reporting time unless prohibited from doing so for reasons determined by management to be beyond the control of the employee.

C. When a leave is for FMLA or for an absence resulting in three (3) consecutive sick days, the employee must provide a medical return to work approval from a medical doctor or other health care professional acceptable to management to return to work.

16.05 DEPARTMENT DIRECTORS

The City Manager shall be authorized to establish different terms for use and accrual of PTO for employees covered by Section 1.02(B).

16.06 CHARGING TIME

PTO will be charged for the time the employee is away from work based on the time clock, if applicable, or in increments of not less than one-quarter (1/4) of an hour.

16.07 UNUSED PTO

- A. Employees may carry over unused PTO hours from one fiscal year to the next to a maximum of 240 hours for employees with less than 5 years of service, 280 hours for employees with more than 5 years but less than 10 years, and 320 hours for employees with more than 10 years of service. Hours in excess of these maximums at the end of the fiscal year will be forfeited. For example: An employee with 36 months (3 years) of continuous service accrues 144 hours in a fiscal year. He/she may carry over their unused balance until they reach 240 hours, excess hours over 240 are forfeited.
- B. Employees working 24 hours shifts may carry over unused PTO hours from one fiscal year to the next to a maximum of 360 hours for employees with less than 5 years of service, 420 hours for employees with more than 5 years but less than 10 years, and 480 hours for employees with more than 10 years of service. Hours in excess of these maximums at the end of the fiscal year will be forfeited. For example: An employee with 36 months (3 years) of continuous service accrues 202 hours in a fiscal year. He/she may carry over their unused balance until they reach 336 hours, excess hours over 336 are forfeited.
- C. In the event of an extenuating circumstance as authorized by the immediate supervisor and the Department Director, an employee who would otherwise forfeit PTO hours at fiscal year-end may carry over up to 50% of the applicable at-risk hours, not to exceed 40 hours. The additional hours carried over will be converted to Personal Day(s) hours and are subject to the provisions of Section 15.01 of these Rules, except no payment of any unused balance will be made to the employee and these Personal Day(s) hours must be used within the next fiscal year or will be forfeited.

For the purposes of this section extenuating circumstances include:

- 1. An unusual number of projects or a significant increase in workload during the fiscal year that requires the skills of the employee and the employee's devotion of time.
- 2. An unexpected increase in responsibilities from assuming duties of a vacant position for thirty (30) calendar days or longer in a fiscal year.
- 3. An emergency or natural disaster which the City requires an employee to work when the employee was scheduled for PTO and the employee cannot take the PTO by the end of the fiscal year.

16.08 PAYMENT OF UNUSED PTO

- A. Payment of unused PTO at time of separation:
 - 1. Regular Full Time employees: Subject to subparagraph D below, upon separation from City employment, employees are entitled to compensation for any balance of unused PTO hours to a maximum of 160 hours for employees with less than 5 years of service; 200 hours for employees with more than 5 years but less than 10 years; and 240 hours for employees with more than 10 years of service.
 - 2. Employees working 24 hour shifts: Subject to subparagraph D below, upon separation from City employment, employees are entitled to compensation for any balance of unused PTO hours to a maximum of 240 hours for employees with less than 5 years of service; 300 hours for employees with more than 5 years but less than 10 years; and 360 hours for employees with more than 10 years of service.
- B. Should an employee die while in service, any balance of unused PTO hours will be paid to the designated beneficiary listed on the form for his City provided life insurance at the same rate as employees who separate from City employment. Regular Full-time employees see Section 16.08 A1 and employees working 24-hour shifts see 16.08 A2.
- C. Payment shall be at the employee's base hourly rate at time of employment termination.
- D. An employee terminated for any of the following reasons shall not be entitled to be paid unused PTO hours at the time of separation:
 - 1. Use of official position for personal advantage.
 - Making false claims or intentional misrepresentation in an attempt to obtain sickness or accident benefits, workers' compensation, or any other benefit.

- 3. Concerted curtailment, restriction of production, or interference with work in or about the City's work stations including, but not limited to, instigating, leading, or participating in any walkout, strike, sit down, stand-in, slowdown, or refusal to return to work at the scheduled time for the scheduled shift.
- 4. Possession, use, sale, attempt to sell, or procure illegal controlled substances at any time whether on or off City property or whether on or off duty; and possession, use, sale or attempt to sell or procure alcoholic beverages while on duty, or City property, while operating or riding in or on City equipment.
- 5. Stealing from the City.
- 6. Intentionally causing the City to be found in violation of Federal or State law.

16.09 UNUSED SICK LEAVE BALANCE AS OF AUGUST 19, 2013

Upon implementation of this policy, employees with a sick leave balance as of August 19, 2013 will retain that full balance of which the hours will only be available for use in a "catastrophic" situation. Catastrophic will be defined as any illness lasting longer than five (5) consecutive working days. To receive payment the employee will be required to provide a medical excuse acceptable to the City. When an employee meets the catastrophic requirements to use unused sick leave, the City will restore the PTO hours used and charge the hours to the unused sick leave balance for the duration of the illness or until the sick leave balance is exhausted. Upon separation of employment, or death while in service, those hours will be paid in a lump sum payment (less statutory deductions) as outlined below using the employee's years of service at date of separation.

Years of Service	% of Sick Leave Payment
0-9 years	0%
10-14 years	30%
15-19 years	40%
20+ years	50%

SECTION 17

BEREAVEMENT LEAVE

17.01 BEREAVEMENT LEAVE

- A. Approved leave in the event of the death of a member of a regular full-time employee's Immediate Family as defined in Section 2 Definition of Terms will be granted as provided below:
 - 1. Up to twenty-four (24) hours with pay if the funeral is in Florida and up to forty (40) hours or twenty-four (24) hours for 24 hours shift employees with pay if the funeral is outside Florida.
 - 2. The minimum leave under this section shall be four (4) hours.
 - 3. Regular part-time employees shall be entitled to one-half (1/2) of the bereavement leave provided to regular full-time employee.
- B. The employee may be required to provide the Director with proof satisfactory to him of death in the employee's immediate family and that the employee attended the funeral before compensation will be approved.
- C. If, in the opinion of the Director, additional days are necessary, accrued PTO may be used, or the employee may be given additional time off without pay.

SECTION 18*

FAMILY MEDICAL LEAVE POLICY

18.01 ELIGIBILITY AND REASONS

Employees, who have worked for the City for at least twelve (12) months, and for at least 1,250 hours during the preceding twelve (12) months and who work at a location where the City has at least 50 employees within 75 miles, may:

- A. Take up to twelve (12) weeks of unpaid leave in a twelve (12) month period for the following reasons:
 - 1. The birth of the employee's child and to care for the newborn child;
 - 2. The placement of a child with the employee for adoption or foster care;
 - 3. In order to care for the employee's spouse, declared domestic partner, child or parent who has a serious health condition;
 - 4. Because of a serious health condition which renders the employee unable to perform the essential functions of the employee's position.
 - 5. Because of a qualifying exigency (as defined below) arising out of the fact that the employee's spouse, child (of any age) or parent is a retired or reservist member of the military on active duty or has been notified of an impending call or order to active duty in support of a contingency operation. Leave under this subsection is not available for an employee whose family member is on active duty as a member of the Regular Armed Forces.
- B. Take up to twenty-six (26) weeks of unpaid leave in a single twelve (12) month period in order to care for the employee's spouse, domestic partner, child (of any age), parent or next of kin who is a military service member who is undergoing medical treatment, recuperation, or therapy, or who is in outpatient status, or who is otherwise on the temporary disability retired list, for a serious injury or illness incurred in the line of duty while on active duty in the Armed Forces.

18.02 DEFINITIONS

The following definitions apply for purposes of this Policy:

A. Serious Health Condition -- A "serious health condition" means an illness, injury, impairment, or physical or mental condition that involves either: (1) inpatient care (i.e., an overnight stay) in a hospital, hospice, or residential medical care facility, including any period of incapacity (i.e., the inability to work, attend school or perform other regular daily activities), or any

- subsequent treatment in connection with the inpatient care; or (2) continuing treatment by health care provider, as defined by the FMLA and the pertinent regulations.
- B. Serious injury or illness A "serious injury or illness" means an injury or illness incurred by a military service member in the line of duty on active duty that may render the service member medically unfit to perform the duties of his or her office, grade, rank or rating.
- C. Child Except as otherwise noted in this policy, "child" means a biological, adopted or foster child; a stepchild; a legal ward; or a child of a person standing in loco parentis (in the place of a parent) and who is either under the age of eighteen (18) or, if older than the age of eighteen (18), is incapable of self-care because of a mental or physical disability.
- D. Parent Parent means a biological, adoptive, step or foster parent, or any other individual who stood in loco parentis (in the place of a parent) to the employee when the employee was a child. Parent does not include parents "in law."
- E. Next of Kin The "next of kin" of a military service member means the nearest blood relative other than the service member's spouse, parent or child, in the following order of priority (unless the service member has specifically designated in writing another blood relative as his nearest blood relative for purposes of military caregiver): (1) blood relatives who have been granted legal custody of the service member, (2) brothers and sisters, (3) grandparents, (4) aunts and uncles and (5) first cousins.
- F. Qualifying exigency A "qualifying exigency" includes leave taken for any of the following reasons: (1) to address any issue resulting from an impending call to active duty deployment on less than seven days' notice, (2) to attend military events and related activities (such as a military ceremony, briefing, family support program, etc.), (3) to make arrangements relating to childcare and school activities, (4) to make financial and legal arrangements, (5) to attend counseling, (6) to spend time with a covered military member who is on a short-term, temporary rest and recuperation leave during the period of deployment, (7) to attend post-deployment activities (such as a military ceremony, event, reintegration briefing, etc.), and (8) any other exigency agreed upon by the City and employee.

18.03 MEASURING THE TWELVE-MONTH PERIOD AND COUNTING FMLA LEAVE

A. For leave taken for any of the reasons listed in Section 1(A), the twelve (12) month period in which eligible employees may take twelve (12) weeks of leave will be calculated using a "rolling" twelve month period measured backward from the date an employee uses any FMLA leave. At any time when a need for FMLA leave arises, the amount of FMLA that an employee would be entitled to use is measured by counting how much FMLA leave the

employee has used during the prior twelve months. If during that prior twelve-month period the employee has already used 12 weeks of FMLA leave, the leave is exhausted. If the employee has not used twelve weeks of FMLA leave during the prior twelve-month period, he or she is entitled to the balance of the twelve weeks that has not been used. For example, if an employee used four weeks of FMLA leave beginning 2/1/11, four weeks beginning 6/1/11, and four weeks beginning 12/1/11, the employee would not be entitled to any additional leave until 2/1/12. However, beginning on 2/1/12, the employee would be entitled to four weeks of leave, on 6/1/12 the employee would be entitled to an additional four weeks of leave, etc.

- B. For leave taken for the reason listed in Section 1(B), the single twelve (12) month period for calculating leave needed to care for a military service member begins when the employee first starts taking leave for that reason and ends twelve (12) months after that date. Leave under Section 1(B) may not exceed twenty-six (26) weeks in any single twelve (12) month period when combined with other FMLA-qualifying leave under any section of this policy.
- C. For leave taken for the birth of a child or placement of a child for adoption or foster care, the entitlement to leave under this policy expires twelve (12) months from the date of the child's birth or placement.
- D. If both spouses/domestic partners work for the City, the combined leave shall not exceed twelve (12) weeks in a twelve (12) month period if the leave is taken for the birth of the employee's child, or to care for the child after birth, for the placement of a child with the employee for adoption or foster care, or to care for the employee's parent with a serious health condition.
- E. If both spouses/domestic partners work for the City the combined leave shall not exceed twenty-six (26) weeks of leave during the single twelve (12) month period described in Section 3(B) above if the leave is taken for the birth of the employee's child, or to care for the child after birth, for the placement of a child with the employee for adoption or foster care, to care for the employee's parent with a serious health condition, or to care for a service member with a serious injury or illness.
- F. To the extent allowed by law, in the event an absence is for a reason covered by this policy, the City reserves the right to count it as FMLA leave whether the employee has requested FMLA leave or not. Leaves covered by workers' compensation and/or a disability plan will also be counted as FMLA leave to the extent the leave qualifies under this policy.

18.04 INTERMITTENT LEAVE OR LEAVE ON A REDUCED SCHEDULE BASIS

A. In the case of leave based upon a serious health condition or a service member's serious injury or illness, leave may be taken intermittently or on a reduced schedule basis, but only if such leave is medically necessary and

the medical need can be best accommodated by intermittent leave or a reduced schedule. If intermittent leave or leave on a reduced hour's basis is required for planned medical treatment, the employee is required to make reasonable efforts to schedule the treatment so as not to unduly disrupt the City operations.

- B. In the case of leave for the birth or placement of a child in adoption or foster care, intermittent leave or working a reduced schedule is not permitted unless the City, in its sole discretion, elects to allow it.
- C. In the case of leave based upon a qualifying exigency, leave may be taken intermittently or on a reduced schedule basis.
- D. If intermittent leave or leave on a reduced hours leave is required or provided, the City may, in its sole discretion, temporarily transfer the employee to another position for which the employee is qualified with equivalent pay and benefits that better accommodates that type of leave.

18.05 EMPLOYEE NOTICE AND CERTIFICATION REQUIREMENTS

- A. For leave that is foreseeable, the employee must provide the City with at least thirty (30) days' notice. If the need for leave is not foreseeable, the employee is required to provide the City with as much notice as is practicable once the need for leave becomes known. Requests for leave should be on approved forms which are available from HR.
- B. The City will require that leave based upon a serious health condition, or a service member's serious injury or illness, be supported by a medical certification from a health care provider. In accordance with applicable regulations, the City may request, at the City's expense, a second opinion from a health care provider of the City's choice (as well as a third opinion if the second opinion conflicts with the first opinion). The City will require that medical certification be submitted showing that a request for intermittent leave or leave on a reduced schedule basis is medically necessary.
- C. The City may require subsequent medical recertification of an ongoing condition from the employee's health care provider every six (6) months in conjunction with an absence, or more often to the extent permitted by applicable law.
- D. The City will require that leave based upon a qualifying exigency also be supported by a certification and supporting documentation, including a copy of the military member's active duty orders or other similar documentation.
- E. Certification forms to be completed under this section are available from HR. If an employee's certification or recertification is deemed by the HR to be incomplete, the HR Administrator will notify the employee of the deficiency and the employee will be provided seven (7) days to cure the deficiency. A failure to complete the certification may result in the denial of leave for the period of time until the completed certification is submitted.

- F. During leaves under this policy, the employee must periodically report on their medical status and intent to return to work. Upon taking such leave, the employee will be advised of the reporting requirements.
- G. For leave taken because of the employee's own serious health condition, the employee is required to furnish a medical certification from his or her health care provider advising that the employee is able to safely resume performing the essential functions of his or her position before the employee will be allowed to return to work.

18.06 HEALTH INSURANCE PREMIUMS

- A. During leaves of absence under this policy, the City will continue to pay its portion of the health insurance premiums and maintain the employee's coverage under the health plan in the same manner as if the employee had been continuously employed during the entire leave period, provided the employee continues to pay his or her share of the premiums.
- B. Should the employee fail to continue to pay his or her share of the premium, notices of proposed insurance cancellation and the opportunity to pay the premium as required by the FMLA will be provided before the cancellation.
- C. The employee will be advised in advance of any changes in premiums so that he or she will have ample opportunity to make arrangements to continue to pay his or her share of the premiums during the FMLA leave.
- D. If the employee does not return to work after the expiration of the leave, the employee will be required to reimburse the City for the employee portion of health insurance premiums during the family leave as permitted by law, unless the employee does not return due to a serious health condition which prevents the employee from performing his job or circumstances beyond the control of the employee. To avoid required reimbursement, appropriate certification from a health care provider may be required if the employee does not return to work because of a serious health condition.

18.07 ACCRUAL

During any period of leave under this policy, accrual of employment benefits, such as PTO and seniority, etc., shall continue. Pension benefits will be determined in accordance with applicable regulations, but employees will be required to make the pension contributions required, if any, under any City sponsored pension plan covering them. Employment benefits to which an employee may be entitled on the day on which the FMLA leave of absence begins will not be lost because of such leave, except for those paid leave days substituted for unpaid leave taken under this policy as described below. Upon return from FMLA leave, employees are entitled to any changes in benefit plans not dependent upon seniority or accrual during the leave period.

Employees will not be disqualified from bonuses based upon attendance or safety for which they qualified prior to leave because of the taking of FMLA leave.

18.08 SUBSTITUTION OF ACCRUED PAID LEAVE

- A. FMLA leave is generally unpaid. However, the City requires employees to use any accrued paid leave (including PTO, personal, etc.) that he or she may have while on FMLA. This means that the employee's FMLA leave under this policy will run concurrently with the use of any accrued paid leave. The employee will be notified of the designation when the leave begins.
- B. Where the leave is not unpaid but the employee is not receiving his or her full pay (such as when the FMLA is covered also by workers' compensation benefits or the employee is receiving compensation pursuant to a disability plan of insurance), accrued paid leave may be used to supplement the employee's pay to bring him or her up to their full salary, to the extent that both the City and the employee agree.

18.09 JOB RESTORATION UPON RETURN FROM FMLA LEAVE

With the exception of certain key employees, employees who return to work from FMLA leave of absence within or on the business day following the expiration of the leave are entitled to return to their job or an equivalent position with equivalent benefits, pay and other terms and conditions of employment. Designation of key employee status and whether such status will affect the employee's right to reinstatement will be made at the time the employee requests leave, or at the commencement of leave, whichever is earlier, or as soon as practicable thereafter if such determination cannot be made at that time.

18.10 FAILURE TO COOPERATE

Employees who fail to provide information to, or otherwise cooperate with, the City in administering this policy, or who provide intentionally untruthful information as to the facts upon which the FMLA leave was granted, may have their leave delayed and/or be subject to discipline up to and including discharge as permitted by law.

SECTION 19

COURT/WITNESS LEAVE

19.01 WITNESS LEAVE FOR THE CITY*

Employees who appear as witnesses on behalf of the City in any judicial or administrative proceeding, or who are subpoenaed as a witness in a matter which involves City business in which the City is not a party in any judicial or administrative proceeding, including deposition, or who are directed by the City to testify in any proceeding shall have all such time treated as compensable work time.

19.02 OTHER COURT-RELATED LEAVE

- A. Employees who become plaintiffs or defendants in personal litigation or who testify or appear on behalf of parties and other persons except the City are not eligible for leave with pay unless they request and are approved for PTO or personal days under Sections 15 or 16.
- B. Employees subpoenaed by the Federal Government or Pinellas County State Attorney's Office as witnesses shall receive their normal pay less any witness fees received.

19.03 JURY DUTY

The City shall make up the difference between a regular full-time employees' pay for his normal schedule provided the employee:

- A. Advises his Director no later than three (3) working days before he is to report for jury duty or when he is first advised, whichever first occurs.
- B. Returns to duty each day he is released from jury duty when two (2) or more hours remain on his scheduled workday or shift unless he gets permission from his Director not to return.
- C. Provides the City with a copy of his check for jury pay.
- D. An employee who attends court or serves jury duty under the conditions described in Section 19.01 above while on PTO shall be allowed to reinstate PTO hours served in court providing satisfactory evidence of the time served on such duty is presented to the City.

19.04 RETURN TO WORK

Employees who attend court or any other legal or administrative proceeding for only a portion of a regularly scheduled workday are expected to report to their supervisor when excused or released.

19.05 REPORTING

An employee subpoenaed to attend court, give a deposition, attend any administrative hearing, or serve jury duty shall promptly notify his immediate supervisor so that arrangements can be made for his absence.

SECTION 20

LEAVE OF ABSENCE WITHOUT PAY (Other than FMLA or in the event that the employee does not qualify for FMLA)

20.01 LEAVE OF ABSENCE WITHOUT PAY

- A. A regular full-time employee may be granted leave of absence without pay for a period not to exceed one (1) year (inclusive of FMLA leave) for sickness, disability or other good and sufficient reasons which are considered to be in the best interest of the City. Such leave shall require the prior approval of the Department Director and the City Manager. Before requesting a leave of absence without pay, an employee must have exhausted all available PTO.
- B. If for any reason the leave of absence without pay is granted, such leave may subsequently be withdrawn, and the employee recalled to service if determined to be operationally necessary by the City.
- C. All employees on leave of absence without pay are subject to these rules.
 - Subject to applicable law, leave without pay shall be granted only when the City determines it will not adversely affect the interests of the City.
 - Failure of an employee to return to work upon expiration of approved leave should be considered job abandonment from the City, absent any unforeseen circumstances as determined by the City Manager.
 - 3. An employee granted a leave of absence without pay, and who wishes to return before the leave period has expired, must make a request to return early to his Department Director as soon as possible to discuss the possible return to work.
 - 4. No PTO, or holiday pay will be accrued or earned by an employee for the time that the employee is on leave without pay.
 - 5. An employee who obtains employment elsewhere, while on authorized leave of absence without pay, will be terminated by the City unless approval has been obtained in advance from the Department Director and the City Manager.
- D. An employee returning from a leave of absence without pay shall be entitled to employment in the same department and position as when the leave began, providing an opening exists. If no vacancies exist, the employee may be offered a lesser position for which he/she is considered

by the City to be qualified. If no such vacancies exist at the time, the employee may be terminated, or the leave extended at the option of the City.

SECTION 21

EFFECT OF LEAVES ON INSURANCE COVERAGE

21.01 EFFECT OF LEAVES ON INSURANCE COVERAGE

- A. Compensable Leave. The City shall continue the employee's group life, medical, dental and short-term disability insurance during compensable leave of absence provided the employee pays his share of the premium.
- B. Workers' Compensation. The City shall continue the employee's group life, medical, dental and short-term disability insurance during an unpaid leave of absence due to a valid workers' compensation injury or illness, provided the employee pays his share of the premium. If the employee's claim is later determined by law to be invalid, the employee shall reimburse the City for all premiums paid in his behalf during the injury. Failure to repay the City such premium upon demand or under terms agreeable to the City will result in termination of employment, and loss of accumulated sick and vacation leave to the extent necessary to cover the reimbursement. To the extent not fully reimbursed, the City may collect the premiums by any means allowed by law.
- C. Other Non-Compensable Leave. If an employee is on an unpaid leave of any type other than FMLA leave, he shall be responsible to pay the full premium for group life and medical insurance beginning the month in which the leave began. The employee shall be entitled to continue coverage for the period of the leave provided he pays the premiums subject to any restrictions imposed by the insurance carrier.

SECTION 22* MILITARY

A military leave of absence will be provided to employees who are absent from work because of service in the U.S. uniformed services in accordance with federal and state law. In order to be eligible for military leave, advance notice of the need for leave is required, unless military service necessity prevents such notice, or it is otherwise impossible.

SECTION 23*

PROHIBITION OF HARASSMENT POLICY

23.01 POLICY

The City is committed to maintaining a work environment free of harassment based upon sex (including gender), race, national origin, ethnicity, disability, religion, color, age, pregnancy, veteran status, sexual orientation, marital status and any other status protected by law. The City will not tolerate the inappropriate harassment of any of its employees, customers or any other individual who does business with the City. It is the affirmative responsibility of all personnel for maintaining a workplace that is free from harassment and intimidation.

The City is committed to promptly and thoroughly investigating all complaints of harassment as set forth in this policy. If, after a thorough investigation, it is determined that inappropriate harassment has occurred in violation of this policy, immediate and appropriate disciplinary action, up to potential discharge, will be taken to promptly end the harassment. Appropriate follow-up steps will also be taken where necessary to ensure that the harassment ceases and does not reoccur.

23.02 DEFINITIONS AND EXAMPLES

A. Sexual Harassment

Improper harassment includes harassment on the basis of one's sex. Prohibited forms of sexual harassment includes, but is not limited to, unwelcome or offensive sexual advances, requests for sexual favors, and any other unwelcome or offensive physical, verbal or visual conduct of a sexual nature such as:

- 1. Unwelcome sexual propositions;
- 2. Sexual innuendo;
- 3. Sexually suggestive remarks;
- 4. Vulgar or sexually explicit comments, gestures, noises or conduct;
- 5. Sexually oriented kidding, teasing or practical jokes; and
- 6. Physical contact of a sexual nature such as brushing against another's body, pinching, grabbing, rubbing, hugging, poking or patting.
- 7. The publication, to anyone, of documents, objects, text, pictures, or

graphics in the workplace that contains material that is of a sexual nature.

8. Using the computer to access any content that contains material of a sexual nature.

Sexual harassment may occur when the intended target of the conduct is not offended, but others find the conduct to be intimidating, hostile, or offensive.

B. Other Forms of Prohibited Harassment

In addition to inappropriate sexual harassment, the City also prohibits harassment on the basis of gender, race, national origin, ethnicity, disability, religion, color, age, pregnancy, veteran status, sexual orientation, marital status or any other status protected by law. Any verbal or physical conduct of an offensive or harassing nature and which is based upon or directed toward any employee based upon any of these characteristics will not be tolerated. Such prohibited conduct includes, but is not limited to:

- 1. Derogatory, critical, offensive or uncomplimentary jokes, comments, displays, posters, other written materials based upon another's gender, race, national origin, disability, religion, color, age, pregnancy and marital status.
- 2. Any physical conduct taken against another individual because of his or her gender, race, national origin, disability, religion, color, age, pregnancy and marital status.
- 3. Teasing or making fun of another individual's ethnicity, accent, cultural or religious beliefs or practices, mental or physical disabilities or medical limitations and other similar characteristics.

23.03 THE PREVENTION OF HARASSMENT IN THE WORKPLACE IS EVERY EMPLOYEE'S RESPONSIBILITY.

All personnel are responsible for maintaining a workplace that is free of harassment and intimidation. If any person experiences or witness's harassment in the workplace, he or she has an affirmative obligation to report the harassment to his or her supervisor or the HR Administrator. If, after an investigation is conducted, it is determined that harassment has occurred, and it is determined that any employee(s) failed to fulfill their affirmative obligation to report it, such a failure may be grounds for discipline.

Upon receiving a report of harassment, supervisors are responsible for immediately reporting it to HR, without regard to whether the harassment involves the supervisor's subordinate employee(s). Failure to report will be

grounds for discipline.

Additionally, supervisors, together with the HR, are responsible for ensuring that their employees are properly trained on the City's policies concerning prohibited harassment and retaliation.

23.04 COMPLAINT AND INVESTIGATION PROCEDURE FOR PROHIBITED HARASSMENT

A. Reporting Prohibited Harassment

Any employee who believes he or she has been the subject of harassment prohibited by this policy must immediately report the harassment to either (1) his or her supervisor; or, (2) the HR Administrator.

Employees are not expected to report harassment to the person they believe is harassing them. However, in such cases, employees are expected to report the harassment to another employee identified above. In the event that the harassment involves the employee's supervisor or the HR Administrator, the employee must immediately report the harassment to their Director or the City Manager.

If, after reporting the harassment as outlined above, the harassment continues or any further incidents of inappropriate behavior occur, employees must immediately report it. Since the City may not be aware that the harassment is ongoing or that its initial handling of the matter has not satisfactorily resolved the issues or caused the offending conduct to cease, employees are required to report any continuing harassment or new incidents of misconduct even where he or she has previously reported a complaint.

B. Sensitivity and Timeliness of Investigation

All complaints will be handled in a timely and sensitive manner. All investigations will be completed as promptly as the circumstances of the allegations allow.

C. Nature of Investigation and Prohibition of Retaliation

Investigation of a complaint will normally include conferring with the parties involved and any named or apparent witnesses. The investigation may also entail reviewing pertinent documents, e-mail communications, pictures and/or any other relevant physical evidence.

All persons who participate in such an investigation shall be protected from coercion, intimidation, retaliation, interference or discrimination for filing a complaint or participating in an investigation. Employees who believe they have been subjected to retaliation on the basis of having filed a report of harassment or having participated in an investigation, must

immediately report it pursuant to Section 23.04(A) above. Any employee determined to have retaliated against another individual in violation of this policy will be subject to discipline up to and including potential discharge.

D. Conclusion of Investigation

All investigations will be conducted in a fair and impartial manner. If, after a thorough investigation, it is determined that prohibited harassment or retaliation has occurred, immediate and appropriate action will be taken to promptly remedy any improper conduct and to ensure that no prohibited actions occur in the future. Such action may include discipline of anyone determined to be in violation of this policy, remedial training concerning the City's policies and procedures relating to prohibited harassment and retaliation, and any other measure determined to be necessary for the effective enforcement of this policy up to and including termination.

23.05 INTERNAL INVESTIGATION PROCEDURE

The following is the procedure for investigating and resolving claims of harassment or misconduct prohibited by this policy.

- A. All incidents, reports or complaints of harassment or misconduct should be filed or reported to the HR Administrator immediately. All such complaints, once filed, shall be investigated and handled exclusively by the City Manager in consultation with the City's Attorney or his designee, unless otherwise specifically provided in this procedure. However, an employee disciplined as a result of a finding of harassment or misconduct or filing a willful and intentional bad faith claim of harassment or misconduct shall be entitled to file a grievance in accordance with applicable policy or an applicable collective bargaining agreement.
- B. Because of the extremely sensitive nature of these types of problems and the potential "spill-over effect" on the job future and personal lives of all involved, the following procedures are established:
 - 1. When a complaint of harassment is made against an employee or person doing business with the City a complaint file will be opened by the HR Administrator which will be separate from the personnel file of the employee involved. Said file will be confidential to the extent allowed by Florida law and will be available only to the City Manager for the specific case, and:
 - a. The City Attorney or his designee;
 - b. Investigators, if any (until the investigation is completed)

assigned the specific case by the City Manager.

- 2. Investigators assigned to handling the complaint will not discuss any aspect of the investigation with anyone, except those persons who have access to the file as provided in Paragraph 1 above and then only upon request or as necessary to complete the investigation.
- 3. The investigators will be charged with the responsibility of gathering information relevant to the complaint. The identity of persons interviewed will remain confidential except to those persons enumerated in Paragraph 1 above unless otherwise directed by the City Manager or required by law.
- 4. Unless otherwise directed by the City Manager, the investigation will be supervised by the City Attorney or his designee.
- 5. Upon completion of the investigation the investigators will make a verbal report to the City Manager.
 - a. A record of the decision of management will be made in the separate file by the City Manager and the parties will be advised of the decision.
 - b. If the decision involved discipline of any employee other than oral reprimand (e.g., written reprimand, suspension or discharge) a copy of the actual notice of disciplinary action will be placed in the employee's personnel file.
 - c. If the person against whom the claim was made is exonerated or no disciplinary action other than oral reprimand is taken, no entry will be made in the personnel file. In such instance, the complaint investigation shall clearly state whether there was no merit to the complaint or that it was insufficiently serious to warrant discipline more severe than oral reprimand. If any oral reprimand was given, the City Manager shall prepare a memo explaining the oral reprimand. The complaint investigation file will not be disclosed to anyone other than as provided in Paragraph 1 above unless disclosure is required by law.

SECTION 24*

PENSION

24.01 SECTION 457 DEFERRED COMPENSATION PLAN

The City maintains a Section 457 Deferred Compensation Plan for eligible employees.

24.02 (LOCAL ORDINANCE) PENSION PLAN

- A. The City maintains a Defined Benefit Pension Plan for eligible employees. Participation by eligible employees is mandatory, and the employee and City share in contributions. The City will advise as to the mandatory City and employee contribution rates.
- B. The City maintains a Section 401a Defined Contribution Plan for MAPS employees. The City only contributes to this plan.
- C. The City maintains Section 401a Defined Contribution Plan(s) for eligible General employees. The City and the employee share in the contributions.

24.03 DETAILS AND COPIES

Copies of the Summary Plan Description, which includes the requirements for eligibility for each of the plans generally described in Sections 24.01 and 24.02 above, have been provided to all eligible employees. Additional copies, as well as a copy of each entire plan, are available in HR. The Summary Plan Description provides details as to eligibility, cost, benefits, and all other related matters.

SECTION 25

PERFORMANCE EVALUATIONS

25.01 THE PERFORMANCE EVALUATION SYSTEM

- A. The process for rating the work performance and work attitude of employees accurately is important to the efficient operation of the City and the job opportunities for employees.
- B. Performance Evaluations (PEs) are given to evaluate the employee's work attitude, quality, quantity and performance and to assist the employee and management in recognizing strengths and weaknesses and to assist in improving areas identified as weak.
- C. PEs will be conducted annually for regular employees and on other occasions as determined necessary by the Director.

25.02 USE

In addition to assisting employees and management in achieving and maintaining acceptable or better job performance, PEs are considered in determining advancement, merit pay increases, disciplinary actions and other job-related actions.

25.03 REVIEW

A regular full-time employee may seek review of his PE under Section 13, provided the decision of the Director shall be final unless the employee alleges a violation of Section 1.05.

25.04 MAINTENANCE OF RECORDS

The HR Administrator will maintain the original copy of completed performance evaluations in each employee's personnel file.

SECTION 26*

DRUG-FREE WORKPLACE AND ALCOHOL POLICY

26.01 POLICY

The City's Drug-free Workplace Policy is aimed at insuring "0" tolerance to illegal drugs at all times and its Alcohol-Free Policy to "0" tolerance under circumstances that affect or might affect the safety and well-being of employees, citizens and others, or the effective operation of City business. In addition, all employees required to have a Commercial Driver's License (CDL) under Chapter 49 CFR Part 383 are subject to controlled substance and alcohol testing rules established by the Federal Highway Administration (FHWA) under the Omnibus Transportation Employee Testing Act of 1991 (revised February 1994), in accordance with 49 CFR, Parts 40, 383, 392, 4, and 392.5 regulatory penalties for infractions are in addition to disciplinary action including termination of employment.

26.02 PROHIBITIONS

- A. *Illegal controlled substances*. The City prohibits the use, distribution, possession, manufacture, cultivation, sale or attempt to sell or distribute illegal controlled substances at any time whether on or off duty, whether on or off City property. Illegal controlled substances are defined by applicable State and Federal laws.
- B. Alcohol abuse. Employees of the City are prohibited from using or possessing alcohol while on duty; while on City premises; while driving a City vehicle, operating a piece of City equipment, or being transported in City vehicles at any time; reporting to work under the influence of alcohol; or, from otherwise using alcohol in a manner at any time which adversely affects the business interests of the City.

26.03 USE OF LEGAL DRUGS

The use of legal drugs, that is drugs prescribed by licensed physicians for a specific medical purpose, is often necessary. However, such drugs can and often do have a direct impact on the vigilance, judgment and/or coordination of the employee and adversely affect the employee's job performance and the employee's ability to work in a safe and efficient manner. This is particularly true in safety-sensitive assignments involving the operation of motor vehicles and other moving equipment. Therefore, an employee for whom a licensed physician or dentist prescribes a controlled substance, must advise the supervisor immediately in order that an evaluation can be made on the impact, if any, on the safe and efficient operation of the City.

26.04 TESTING

- A. Substances Tested For: Employees will be subject to drug testing for the detection of the following illegal drugs/drug groups, as well as others that may from time to time be declared illegal by state or federal law:
 - 1. Amphetamines
 - 2. Barbiturates
 - 3. Benzodiazepines
 - 4. Cannabinoids (marijuana)
 - 5. Cocaine
 - Methadone
 - 7. Methaqualone
 - 8. Opiates (heroin, morphine, codeine)
 - 9. Phencyclidine (pcp)
 - 10. Propoxyphene
- B. Testing for Illegal Controlled Substances Classes of Employees/Circumstances, subject to applicable law:
 - 1. Employees in special risk and safety sensitive positions.
 - a. Special risk and safety sensitive employees include all employees in all classifications of requiring a CDL license, and firefighter EMTs and firefighter Paramedics regardless of their rank. Other employees who are considered special risk or safety sensitive shall be notified of said status in writing.
 - b. Candidates and employees are subject to testing on the same basis as other employees under Section 26.04(B)(2), except no reasonable suspicion is required for testing such employees for illegal controlled substances for:
 - (1) Any accident involving any personal injury that results in a worker's compensation claim or serious damage to property occurs.
 - (2) As otherwise allowed or required by law, provided unless required by law, there shall be no random drug testing

except in conjunction with rehabilitation under Section 26.08.

- 2. Non-Safety Sensitive/Special Risk Employees.
 - a. When an employee is involved at any time directly in an equipment or vehicular work-related accident, any accident on-the-job, or in any unsafe and/or negligent maintenance or operation of the City's equipment or vehicles at any time where in the opinion of the City Manager the employee was at fault or the employee's conduct contributed to the accident and there is reasonable suspicion to believe the employee was in violation of Section 26.02(A) or (B).
 - b. When reasonable suspicion exists to believe the employee is using drugs or alcohol in violation of this policy. Reasonable suspicion is a belief by two (2) or more supervisors or managers that an employee is using or has used drugs or alcohol in violation of this policy drawn from specific objective and articulable facts and reasonable inferences drawn from those facts in light of experience. Among other things, such facts and inferences may be based upon:
 - (1) Observable phenomena while at work, such as direct observation of drug use or of physical symptoms or manifestation of being under the influence of a drug or alcohol:
 - (2) Abnormal conduct or erratic behavior while at work or a significant deterioration in work performance;
 - (3) A report of drug use;
 - (4) Evidence that an individual has tampered with a drug test during his employment with the City;
 - (5) Information that an employee has caused, contributed to, or been involved in an accident while at work;
 - (6) Evidence that an employee has used, possessed, manufactured, cultivated, sold, solicited, or transferred drugs;
 - (7) Frequent absences from work without a satisfactory explanation.

- C. Employee Rights When testing to determine the presence of illegal controlled substances under subparagraphs (1) and (2) above:
 - 1. Employees and job candidates have the right to consult with the testing laboratory for technical information regarding prescription and non-prescription medications. The name, address and telephone number of the testing laboratory will be provided to the employee or job applicant upon request.
 - 2. All test results will be kept confidential and will only be provided to managerial employees on a need-to-know basis.
 - 3. For tests under Section 26.04(B)(1)(b)(1) and (2), Employer shall meet with and inform an employee that, in the opinion of the Employer, there is a basis for reasonable suspicion and of the Employer's intention to schedule a drug or alcohol screen or test. At said meeting, the Employer shall consider the comments of the employee regarding the matter and shall then make a final determination of whether to proceed and require the screen or test.
 - 4. Employee may upon his request have a representative present at said meeting; however, the meeting shall not be delayed because the employee wishes to have a specific representative present. If it is determined by the Employer that a drug or alcohol screen or test will be required, the employee shall be immediately escorted to the appropriate facility for the test. Refusal by the employee to submit to said test may be grounds for disciplinary action, including termination of employment.
 - 5. If the employee is in a collective bargaining unit, the representative in subparagraph (4) above may be a Union representative.
 - 6. Procedures for testing for the presence of illegal controlled substances shall be conducted consistent with the provisions of *Florida Statute* 440.102(5)(a) through (o) and (6) for alcohol, a positive result is 0.02, or greater. For drugs, a positive result is in accordance with the detection levels established by HRS guidelines.
 - 7. The common and chemical names of the substances identified in Sub-section A above, a copy of *Florida Statute* 440.102(5) and (6) and a list of local drug rehabilitation programs is available from HR.

26.05 REPORTING AND CONVICTION OF ALLEGED CRIMES INCLUDING DRUGS OR ALCOHOL

A. All employees must report to their supervisor any arrest, indictment or conviction of a drug or alcohol related violation or alleged violation of law

not later than the next workday after they become aware of it. Failure to so report may result in immediate termination.

- B. Upon conviction of a crime involving illegal drugs, the employee will be immediately terminated.
- C. Without regard to prosecution or conviction by appropriate governmental entities, the City may, at its option, conduct its own independent investigation to determine whether or not there has been a violation of the City's drug and/or alcohol policy. If, in the opinion of the City, it believes a violation has occurred, it will take whatever disciplinary action it deems appropriate regardless of the ultimate outcome of any criminal case that may be brought against the employee.

26.06 DISCIPLINE FOR VIOLATION OF POLICY

Employees who violate this policy or who are directed to take a physical examination, blood, breathalyzer, urinalysis or other test allowed by law, and refuse or fail to do so when and as directed; or who, after having taken such examination and/or test are determined to have utilized illegal controlled substance at any time or to have violated the City's Alcohol Abuse Policy, shall be subject to immediate termination; provided, however, if the presence of an illegal controlled substance is established as a result of the test, the employee may, within five (5) working days of receipt of written notification of a positive result, request an opportunity to explain the result to the City and/or the medical review officer.

26.07 EMPLOYEE INJURED ON THE JOB

Any employee injured on the job who refuses to submit to a drug test, or has a positive confirmation test, in addition to other provision of the policy, may forfeit his eligibility for all workers' compensation medical and indemnity benefits depending on applicable law.

26.08 EMPLOYEE ASSISTANCE PROGRAM

The City has an Employee Assistance Program (EAP) with one of its missions being to assist employees who voluntarily report drug or alcohol related problems, which have not yet adversely affected their job or City operations. The City may require any employee in violation of this policy, whether he voluntarily reports his problem or not, to participate in the EAP or other medical and rehabilitative assistance programs as a condition for continued employment. For further information regarding the EAP, contact the HR Office.

A. Employees Who Voluntarily Ask for Help. Employees with drug or alcohol related problems who wish assistance through the EAP may contact the EAP provider on a confidential basis or through the HR Office. If the request is made through the HR Office, City referrals will be made only upon execution by the employee of a release to the EAP provider to keep HR advised as to the employee's attendance and progress in the

rehabilitation program. If the employee has a satisfactory performance record and is otherwise qualified to perform his job, the City may grant the employee an unpaid leave of absence for a period determined by the City to participate in a City approved treatment or rehabilitative program. Such a leave will be granted only one (1) time. This employee will be responsible for all expenses resulting from the treatment or program to the extent they are not covered by insurance.

- B. Other Employees. In the event the City discovers a violation of this drug or alcohol policy, or an alcohol-related problem that adversely affects or may adversely affect the employee's performance or the City business, the City may proceed to discipline the employee up to and including discharge, or at its option, require the employee to undergo approved medical or rehabilitative assistance. The City may grant the employee leave with or without pay to participate in a rehabilitation program, including referral to the City EAP program. Such leave may be granted only one (1) time. Allowing of rehabilitation under the City EAP program will be conditioned on the execution of a consent by the employee to allow the EAP provider, or persons providing medical or rehabilitative assistance to keep HR advised of the employee's attendance and the success of rehabilitation. The employee will be responsible for all expenses resulting from the treatment or rehabilitation to the extent they are not covered by insurance.
- C. Return to work. Employees who are granted a leave of absence under paragraph A or B above must successfully complete all EAP, medical and other rehabilitative requirements established by the City within a reasonable amount of time. A successfully rehabilitated employee who has been granted a leave of absence under A above shall be returned to his former job provided he successfully completed rehabilitation within the period of his leave. Return is also conditioned on maintaining all certifications required of the job. Employees who successfully complete rehabilitation under B above within the period of his leave will be returned to his former job if vacant, but if not to any vacancy which the City considers him qualified to perform, if any, and if there is none he shall be terminated.
- D. Re-testing. Employees allowed to return to work from an illegal controlled substance problem shall be subject to re-testing any time without notice and must submit to such test as and when directed by the City for one (1) year after they have been free of illegal drugs as determined by the City, or its designee.

26.09 REPORTING VIOLATION OF THE POLICY

A. Reporting violations. It is the obligation of every employee of the City to report violations of the City's drug and alcohol abuse policies. Failure to report may subject employees to discipline up to and including termination.

- B. Good faith reports. Any employee who in good faith, based upon reasonable suspicion or observation, reports an alleged violation of these policies, or any supervisory or managerial employee who investigates or take action in good faith based on reasonable suspicion or observation shall not be harassed, retaliated against, or discriminated against in any manner for making reports, participating in the investigation or because of any reasonable action he takes as a result of the investigation.
- C. Bad faith claims. Any knowingly false reporting of a violation of the policies set forth herein shall subject the employee to immediate termination.

26.10 COORDINATION WITH THE HR OFFICE

All action taken by members of management under this Section 26 must be coordinated through the HR Administrator to ensure compliance with all applicable laws.

SECTION 27

EDUCATIONAL REIMBURSEMENT PROGRAM AND INCENTIVE PAY

27.01 TUITION

The City may authorize payment for tuition for courses it considers will be helpful to successful completion of the mission of the City subject to budgetary constraints as follows:

- A. The City Manager, or his designee, shall determine if the course or curriculum is related to the employee's job or contributes to the long-range value of the employee to the City.
- B. If the course is reimbursable through some other source, then provisions of the City's education tuition payment plan shall not apply.
- C. Based on budgetary constraints, the City Commission shall budget a set annual amount for tuition reimbursement. Once that amount is dispersed, no other reimbursement requests will be approved.

27.02 ELIGIBILITY

- A. Full-time employment with the City at least one (1) year, unless the City Manager grants an exception.
- B. The educational training can be at the post high school or adult educational levels.
- C. No more than two (2) courses per quarter or semester (or equivalent period of time) may be reimbursed.
- D. No employee will be reimbursed for more than six (6) classes in a calendar year unless previously approved by the City Manager.

27.03 REQUESTS AND PAYMENT

- A. The employee who wishes to participate in the program must submit a completed application to their Director prior to new fiscal budgets being finalized. In order to process the request in a timely manner, employees should make every effort to request necessary funding as soon as school schedules are released. Failure to obtain approval in advance may result in requests for reimbursement being denied. HR will notify employees when applications are due.
- B. Employees are encouraged to attend any accredited technical or trade school, Florida College or University; however, tuition reimbursement shall be paid at the Public College or University tuition rate (see application). Correspondence schools will not be considered for tuition reimbursement. Employees shall not be reimbursed for mileage or personal expenses unless they are required by the City Manager, or his designee, to take the

course.

- C. The employee shall submit to the HR Office the final grade, certification, or degree (if applicable) and paid receipts in order to receive reimbursement. Students utilizing financial assistance and that have deferred payment until a later date can request payments be made directly to the lender. Failure to disclose FAFSA or other financing for school is a Group II disciplinary offense.
- D. Tuition Reimbursement will be made according to the following matrix:
 - 1. Grade A = 100%
 - 2. Grade B = 75%
 - 3. Grade C = 50%
 - 4. Grade D or below: 0%
 - 5. Grade P = 100%

27.04 REPAYMENT BY THE EMPLOYEE

The employee who receives any funds under this program shall be required to work for the City for at least one (1) year from date money is received or shall reimburse the City for every dollar received upon termination. The City reserves the right to withhold the money owed from any final paycheck upon separation subject only to applicable law. The City Manager may excuse repayment in whole or in part for circumstances he determines to be extenuating and justifiable.

27.05 SCHEDULES

The City will attempt to rearrange work schedules for classes if it does not interrupt the normal workflow.

27.06 INCENTIVE PAY

When the State or other governmental agency mandates additional pay for a class of employees and funds or reimburses the City for same, the City will provide the incentive pay to those employees who meet and maintain the qualifications to receive the incentive pay. Department Directors are required to advise the HR Office of employees in their Department who are eligible for such pay.

SECTION 28*

MISCELLANOUS

28.01 SMOKING/TOBACCO PRODUCTS

- A. The purpose of this policy is to protect the public health, comfort, and environment for citizens and employees by creating areas in public places and at public meetings that are reasonably free from tobacco smoke, to comply with Florida Statute, the Florida Clean Air Act and problems created by the use of other tobacco products in the workplace.
- B. No person may smoke or use any tobacco product in any City building, facility or vehicle. This includes, but is not limited to, private offices, hallways, rest rooms, conference rooms and break rooms, even with only one (1) occupant.
- C. Smoking is permitted outdoors during scheduled work breaks in accordance with Federal, State, County or City ordinance except for the following conditions:
 - Whenever a safety hazard exists;
 - 2. In any area where smoking is specifically prohibited by Federal, State, County or City Ordinance;
 - 3. In any area posted "No Smoking".

For purposes of this policy "smoking" includes, but is not limited to smoking cigarettes, cigars, and pipes, but also vaping, and using ecigarettes or any other form of smokeless tobacco.

- D. The City Manager shall be responsible for ensuring City-wide implementation of this policy. Directors and Managers shall be responsible for uniform implementation of this policy in their respective work areas, facilities and buildings.
- E. Violation of this policy shall subject the employee to disciplinary action up to and including termination.
- F. Citizens, clients, contractors and visitors to City facilities shall be expected to comply with this policy. Violators shall be requested to extinguish their smoking material or to leave the building/facility area if they refuse to do so.

28.02 UNEMPLOYMENT/RE-EMPLOYMENT COMPENSATION

- A. The City is registered with the State of Florida Department of Economic Opportunity Reemployment Assistance (formerly known as Unemployment Compensation)
- B. Terminated employees who file a claim and are determined qualified under applicable law may be eligible to receive reemployment assistance benefits. Employees who are terminated for violation of the City Drug Free Workplace Policy may not be entitled to reemployment assistance benefits under Florida law.

28.03 WORKERS' COMPENSATION

- A. When an employee is injured on the job and loses time from work as a result of that injury, the employee will receive a City subsidy to make up the difference in straight time income for their regular schedule for a period not to exceed three (3) months. If Workers' Compensation continues after three months the employee will receive from the City only that which is required under Workers' Compensation Law. At the request of the employee the balance can be deducted from the individual's unused sick, if applicable, or PTO balances to make up their regular straight time income.
- B. Injury or compensable illness shall be determined to have incurred while on duty with the City only if such injury is a compensable injury under Florida's Worker's Compensation Law.
- C. Length of disability shall be determined by the Employer's physician in accordance with Worker's Compensation Law.
- D. Paid Time Off (PTO) accrual shall continue for a maximum of twelve (12) months for employees who are receiving worker's compensation benefits due to a compensable on-the-job injury or compensable illness, unless placed on leave without pay status.
- E. If the employee's claim is later determined by law to be invalid, the employee shall reimburse the City for supplemental compensation received and as to benefits received or accrued, he shall be treated as having been on unpaid leave of absence; provided, he shall not be required to repay accrued PTO used to supplement workers' compensation. Failure to repay the City upon demand or under terms agreeable to the City will result in termination of employment, and loss of accumulated unused sick leave to the extent necessary to cover the reimbursement. To the extent not fully reimbursed, the City may collect by any means allowed by law.
- F. When an employee is injured on the job, they must first report to their supervisor before being seen at a medical clinic, unless the injury needs immediate medical attention; then 911 should be called. The supervisor

will instruct the employee on how to proceed and assist in completion of necessary form, including the "Notice of Injury" form which must be submitted following an accident or injury, regardless of severity within twenty-four (24) hours. Failure to submit such form could result in disciplinary action.

28.04 LIGHT DUTY

- A. If an employee is released by the workers' compensation physician for "light duty," return to light duty shall be at the option of the City based on its operational needs. Refusal to accept a light-duty assignment by the City Manager which the employee is capable of performing in accordance with applicable law will result in termination of employment. An employee can choose to use their PTO in place of working light-duty.
- B. The employee will be paid his normal hourly rate for light duty work and, if applicable, must clock in and out daily.
- C. When an employee is released from Workers' Compensation to return to full duty, any subsequent absences related to that sickness or injury will not be paid by the City unless the City receives payment from the Workers' Compensation carrier. The employee may utilize accrued PTO to compensate for lost wages so long as the employee meets the requirements for use of PTO under the Personnel Rules and Regulations or any collective bargaining agreement. The three-month limitation set forth in subsection 28.03(A) above shall apply.

28.05 DESKS, LOCKERS, OTHER CITY PROPERTY/EQUIPMENT, PERSONAL PROPERTY

- A. Employees may request a locker if available.
- B. All personal equipment, clothing, property and belongings, including, but not limited to, car trunks and lunch bags, brought onto the City property or placed in City property, as well as lockers, desks, file cabinets and other City equipment used by employees, are subject to inspection by management at any time if management has any reason to suspect they contain or may contain anything the possession of which would violate a City or department ordinance, rule, procedure or policy, or any Federal, State or Local law or regulation.
- C. In the event the employee is unavailable, or it is deemed necessary to access any item described in Section 28.04(B) above without the employee's presence, the Director must request authority to access from the City Manager. The City Manager shall make the determination.
- D. Prescription eyeglasses, contact lenses, hearing aids and watches of employees that are lost, damaged or destroyed in the line of duty, except through employee negligence, shall be replaced or repaired at the Employer's expense subject to the following restrictions. The Employer

shall not be responsible for any other personal property which is lost, damaged or destroyed in the line of duty.

- 1. The maximum reimbursement for prescription eyeglasses, contact lenses and hearing aids shall be the cost of replacing the precise item of one of equal quality or two hundred fifty dollars (\$250.00), whichever is less.
- 2. The maximum reimbursement for watches shall be the cost of replacing the item with one of equal quality or one hundred fifty dollars (\$150.00), whichever is less.
- 3. Request for reimbursement for lost or damages personal property shall be made in writing to the employee's immediate supervisor during the work shift in which the article of personal property was damaged or lost.
- 4. Except when lost, the item for which reimbursement is sought must be turned in along with the written request for reimbursement.
- 5. Reimbursement for lost or damaged personal property must be approved by the City Manager.

28.06 RECORDS AND REPORTS

HR is responsible for establishing and maintaining comprehensive central personnel records of all City employees.

- Α. All records and materials relating to the administration of the City Personnel Management System are the property of the City. Generally, in accordance with Florida's Public Records Laws, an employee's personnel fie is a public record. However, many exemptions exist in the law that may exempt information contained in the personnel fie from being provided in response to a public records request. In addition, there are certain categories of individuals under F.S. 119 whose personal identifying information such as home address, work location, and information about their spouse and children is exempt form disclosure. HR will provide you with the applicable form if you are entitled to the exemption. Request for information will be forwarded to the City Clerk's Office to determine what information may be exempt and for fulfillment of the request. employee's responsibility to advise HR and complete the form if they believe the exemption is applicable. In the event the exemption is challenged it is the employee's obligation to provide all information needed to the City to support the claim of exemption.
- B. Employees should keep their personnel records current. This means immediately notifying HR of any changes; such as, change of address (even if temporary), change of telephone number, change of beneficiary, number of dependents, divorce, marriage or any status change not

previously reported, from that which was originally given at the time of employment. This is the responsibility of the employee and failure to comply may result in loss of employee benefits.

C. HR should be informed of any special training courses completed by an employee. Copies of diplomas or certificates shall be forwarded to the HR Office to become a permanent part of the employee's personnel file.

28.07 EMPLOYEE TRAINING AND DEVELOPMENT

It is the responsibility of the City Manager in conjunction with Department Directors and the HR Administrator, to foster and promote in-service training of employees. The purpose of this training is to improve the level of service rendered to the public, the quality of personnel, and to assist employees in preparing themselves for advancement in City service. Department Directors in cooperation with the HR Administrator will establish standards for training programs, assure that training is carried out as approved, and prepare certificates or other forms of recognition to persons who satisfactorily complete approved courses and programs. The HR Administrator will provide assistance to Department Directors in developing and conducting training to meet specific needs of their departments and to assure that employee, supervisory and management training are available to all Departments.

28.08 DEDUCTIONS

Federal withholding, Social Security and pensions are deducted from paychecks in accordance with law and City ordinance. Voluntary deductions authorized by the City Manager, are made only upon written authorization of the employee and approval by HR and the Finance Department. Employees who believe that an improper pay deduction has been made, may utilize the grievance procedure in Section 13 to file a complaint. If the deduction is found to be improper, the City will reimburse the employee.

28.09 CELL PHONES

When it is deemed that a City employee has a need for a cell phone to effectively perform their job requirements, the following policy shall apply:

- A. A requisition for the cell phone shall be submitted along with written authorization from the Director explaining the reason for the need for the employee to have a cell phone. The cell phones shall remain the property of the City and will remain so until it is declared surplus and disposed of in a manner acceptable to the City. All bills for the use of the cell phone shall be sent to the City, care of the Finance Department. The employee shall reimburse the City for all personal calls transacted by the use of the cell phone at the rate of 5 cents per minute. Reimbursement shall be by cash or by submittal of a check or money order made out to the City.
- B. If an employee is using their personal cell phone for City business calls, they shall be responsible for full payment of their bill and the City will pay

them a stipend of \$25 per month through payroll (\$300 annually).

C. Should a dispute arise involving any of the above procedures, the decision of the City Manager shall be considered final and binding.

28.10 ELECTRONIC COMMUNICATIONS, INCLUDING TEXT MESSAGES, USAGE AND RETENTION

The availability of e-mail and Internet resources are offered to the employees for the furtherance of their work as City employees. However, it is recognized that occasional communication between employees at work via e-mail; for personal reasons under circumstances that do not take away from or interfere with their duties or the duties of those employees with whom they communicate, is not prohibited except:

- A. No e-mail will contain any material which is political, slanderous, controversial, critical of any person or entity, or which contains vulgar language, reference to sexual matters or is otherwise inappropriate.
- B. E-mail for personal reasons should be used only when it is important for the employee to communicate with another employee at work about a personal matter and another method of communication would take time away from the employee's work. E-mail is not to be used for "chit-chat" and telling jokes.
- C. The City Manager will determine when this section is being violated.

E-mail transmissions, e-mail passwords, and any information transmitted via the City e-mail network are the property of the City of St. Pete Beach and are open to examination by the public, news media or any City official. Employees should not have any expectation of privacy with respect to the contents of such communications. The City reserves the right to make public all personal e-mails, therefore, employees should not put anything in an e-mail they do not want to be made public upon a proper request made in accordance with applicable law. E-mail transmissions are governed by Chapter 119 F.S. and will be retained by the City in accordance with that statute. This can be done either by printing a hard copy which will be kept according to the required retention period set by the State of Florida or by backing up to electronic mail and retaining it in digitized form for the required retention period. It shall be the responsibility of the last receiver on an e-mail chain to ensure the proper retention of the information.

E-mail transmissions and web site activities that are vulgar, offensive, obscene, discriminatory, harassing, or otherwise inappropriate are prohibited.

Checking personal e-mail accounts (i.e. AOL, hotmail, etc.) through City computers must be kept to a minimum and not interfere with regular work duties.

Chat room usage is expressly prohibited

E-mail transmissions that unlawfully distribute copyrighted materials, coordinate illegal activities (e.g. gambling) and facilitate illegal conduct are strictly prohibited.

E-mail passwords and corresponding accounts are non-transferable Employees are responsible for maintaining the confidentiality of their passwords and corresponding accounts.

Employees may not install unapproved software packages on their computers without obtaining advance permission from the City Manager.

Violation of this policy will subject the employee to disciplinary action, up to and including termination.

To ensure that the use of the City's electronic communication system is consistent with the City's legitimate business interests and not a violation of this section, authorized representatives of the City will monitor the use of such equipment from time to time. No notice of such monitoring will be given.

REMEMBER: Our computer system is for City business and any personal use should be as limited as possible and must comply with all the standards set forth in this policy.

Text messages regarding city business, whether on a public or private phone, are considered and subject to public records and shall be limited to notifications, scheduling, and other conversations that do not have administrative value. The employee will be responsible for forwarding the message to their city email account if the conversation needs to be retained for administrative value.

28.11 SOCIAL MEDIA POLICY - EMPLOYEES

Purpose: To ensure the proper use of the City's social networking sites and to enhance communication, collaboration, and information exchange; streamline processes; and foster productivity. This policy establishes the City's position on the utility and management of social media and provides guidance on its management, administration, and oversight.

B. Publicly posted information on the City's social media sites will be professional and reflect positively on the City of St. Pete Beach, its employees, boards, programs, policies, and services. This policy provides rules and guidelines regarding the use of the City-provided social media by employees and elected officials and shall not be construed or interpreted in any way to infringe upon a person's right to free speech under the Florida or United States Constitution.

D. Definitions:

 "Social media" means and includes internet technologies that facilitate and promote interactive communication, participation, and collaboration. Examples of social media included, but are not limited

- to, SPB Connect, Facebook, Twitter, Flickr, Vine, Instagram, Blogger, LinkedIn, Tumblr, Foursquare, YouTube, and the interactive tools and functions they provide to users.
- "City Employee" or "Employee" means a person hired and currently employed by the City of St. Pete Beach. The term does not include the Mayor, City Commissioners, City Board or Committee Members. Independent contractors of the City may be authorized only at the City Manager's discretion.
- "City social media sites" means social media sites which the City establishes and maintains, and over which it has control over all postings, except for advertisements or hyperlinks by the social media site's owners, vendors, or partners. City social media sites shall supplement, and not replace, the City's required notices and standard methods of communication.
- E. Overview: The intent of social media is informational purposes only. If a user wishes to contact City Commission, Administration, or to request City services, they should visit the City's official website. The City's website will remain the official online location for content regarding City business, services, and events. Whenever possible, links to more information will direct users back to the City's official website. City employees will check facts, cite sources, acknowledge and correct errors and check spelling and grammar before publishing any posts on City social media sites. Further, employees may be personally responsible for the content they publish on blogs or any other form of user-generated media.
- F. Employees posting content shall comply with this policy. The City reserves the right to update this policy. The City's social media sites will clearly indicate that all content posted is subject to Florida's Public Records Law (Section 119.07, F.S.). Users will be informed that public records requests shall be made by contacting the custodian of records, the City Clerk.
- G. Establishing a City Social Media Account: City employees acting in their official capacities on behalf of their City may only create City social media sites and/or post City information on City social media sites after obtaining the prior approval of the City Manager. Employees who post information or comments in their personal capacity on City social media sites are limited to posting their own opinions and beliefs and not the official views or positions of the City. Where the context of an employee's personal post does not make it clear that the views posted are the employee's views and not the City's, the employee must state in the posting that the views are his or her own and not the views of the City.
- H. Terms of Use of City Social Media Sites: City social media sites are not monitored all the time. Do not use social media if you need to report an emergency situation or a time-sensitive issue. All emergencies should be reported by dialing 911.

- I. City employees and officials may use social media as a function of their employment with the City, when authorized and within their respective delegated areas of employment responsibilities and duties. The City may use the social media forum for one-way communications unless otherwise directed by the City Manager. The City reserves the right to edit or remove content at its discretion and based on the criteria listed below. The City's intent is not to create a public forum, but to maintain a forum related to topics posted by the City, with language appropriate for citizens and the general public, (including minors), to read. Inappropriate content includes:
 - 1. Comments not topically related to the particular article being commented on.
 - 2. Content that promotes, fosters, or perpetuates discrimination or harassment on the basis of race, creed, color, age, religion, gender, marital status, veteran status, status with regard to public assistance, national origin, physical or mental disability or sexual orientation.
 - 3. Solicitations of commerce.
 - 4. Sexual content or links to sexual content.
 - 5. Comments to support or oppose political campaigns or ballot measures.
 - 6. Conduct or encouragement of illegal activity.
 - 7. Information that may tend to compromise the safety or security of the public or public systems.
 - 8. Content posted in violation of copyright, trademark, patent or any other legal ownership interest of any other party. Accusations that a particular individual is guilty of any criminal conduct or immoral activity.
 - 9. Defamatory, false or misleading material.
 - 10. Formal complaints against the City and its employees, officials, and contractors. If a person believes there is a legitimate basis to make a complaint against the City, any of its employees or any third party, then appropriate alternative means of registering the complaint must be utilized.
 - 11. Harassing and threatening comments. The use of City social media sites shall be for the purpose of allowing the City to distribute information regarding topics determined by the City.
- H. Authorized Use: The City Manager is responsible for determining who is

authorized to use social media on behalf of the City.

- The City Manager will determine the appropriate level of employee access, which includes identifying what social media sites, or types of sites, each department or individual is approved to use, as well as define their respective capabilities: publish, edit, comment or view only.
- 2. Any employee authorized to post items on any of the City's social media sites shall review, be familiar with, and comply with the social media site's use policies and terms and conditions.
- 3. Authorized users shall be provided a copy of this Social Media Policy and are required to acknowledge their understanding and acceptance by signing and dating the last page of this document, and returning the original copy to the City Manager's Office
- City use of social media sites will comply with all provisions of Florida law, ordinances, resolutions and policies of the City including, but not limited to, the Florida Public Records Act (Chapter 119, Fla. Stat.) and the Standards of Conduct applicable to Public Officials and Employees (Section 112.313, Fla. Stat.).

28.12 TAKE-HOME VEHICLES

- A. Some City employees have City vehicles at their disposal so they may carry out their duties properly. Because City vehicles are typically marked, the way they are operated has a direct impact on the public image of the City. Employees operating City vehicles will adhere to the following rules:
 - 1. Operator will possess a valid Florida driver's license.
 - 2. All vehicular safety laws will be observed.
 - 3. Safety belts will be used at all times when the vehicle is in operation.
 - 4. In the event that a City vehicle is involved in an accident, the employee will notify the appropriate law enforcement agency and the employee's immediate supervisor, the Director and Administrative Supervisor.
 - 5. City vehicles will be used for official business only; they will not be used for pleasure or personal business unless otherwise approved by the City Manager.
 - 6. Any employee who abuses a City vehicle in any manner will be subject to disciplinary action including possible payment for the

deductible portion of damages to the vehicle.

- B. Overnight use of a non-assigned City vehicle will require approval of the City Manager before a vehicle is driven home. Requesting employee must possess a valid Florida driver license. Such use normally will be authorized only under the following conditions:
 - 1. The employee requires use of a vehicle overnight or whose responsibilities involve an evening or weekend public appearance. (i.e. Code Enforcement)
 - 2. The employee holds a position with duties and responsibilities involving on-call or standby status.
 - 3. The employee is traveling on City business.
- C. In addition to the rules for operating a City vehicle, employees who are assigned overnight use of City vehicles will adhere to the following rules:
 - Under no circumstances will the vehicle be used for transporting members of the employee's family or for anything pertaining to personal business or pleasure unless otherwise approved by the City Manager.
 - 2. The employee will be responsible to keep the vehicle in clean condition.
- D. Vehicle assignments may change from time to time, and each vehicle will be available for other purposes when not being used by the employee to whom it is normally assigned.
- E. All vehicles not being driven home by employees will remain parked overnight in the building parking lot where the vehicle is assigned.

28.13 SAFETY

- A. All Directors, managers and supervisors are to take all reasonable steps to ensure that City employees work in an environment free from hazards and dangerous conditions which they observe or which, with the exercise of reasonable diligence, they should have observed. They also are responsible to enforce all City, Departmental, State, Local and Federal laws relating to safety applicable to their area of responsibility.
- B. All employees are required to abide by all applicable City, Departmental, State, Local and Federal safety laws and regulations applicable to their area of responsibility.
- C. All employees should report to their Department Director or Supervisor,

any condition, equipment or practice they consider to be unsafe.

D. Employees who are required to wear safety equipment and clothing provided by the City must do so. Failure to do so will justify immediate termination.

SECTION 29*

GROUP INSURANCES

29.01 HEALTH AND DENTAL INSURANCE

The Employer shall provide health and dental insurance selected by the Employer to include dependent coverage at the option of the employee. At this time, in the event that more than one (1) plan is offered, the Employer shall pay the individual premium cost for the lowest plan offered for the employee and the employee is responsible for paying any additional individual premium and the employee is responsible for paying the entire premium for dependent coverage. However, from time to time the City subsidizes dependent coverage. To be eligible for employee insurance, an employee must be classified as regular full-time. Those employees with a 30 hour a week schedule shall be provided health insurance.

29.02 DEPENDENT COVERAGE

The employee shall have the choice of plans offered by the Employer and shall be responsible for the entire premium cost for dependent coverage not paid for by the City from time to time.

29.03 LIFE INSURANCE

The Employer will provide life insurance coverage for employees in the amount of one times the employee's annual base salary, to a maximum of \$100,000.00. This coverage will be in addition to any life insurance required by state law, if any, for a particular classification or group of employees.

29.04 RETIREES

Upon retirement under a City sponsored retirement plan, a retiree shall be allowed to continue participation in the City Group Medical Insurance Plan for himself and his dependents at his own cost under conditions acceptable to the City as to payment, subject to the terms of the Plan and applicable law.

29.05 CAROUSEL BENEFITS PROGRAM

- A. An employee may elect not to participate in the City Group Medical Insurance Plan upon proof acceptable to the City that the employee is covered by another health plan.
- B. An employee wishing to opt out of the City Group Medical Insurance Plan shall complete the forms and supply proof of other coverage with the HR Office.
- C. If approved by the City, the employee will receive an amount set the City Manager annually in lieu of Group Medical Insurance coverage for those weeks he would have been covered by the City plan but for his opt out.

- D. The employee may utilize the amount received as he wishes, including participating in voluntary benefit plans offered by the City.
- E. It is the obligation of the employee to advise the City in writing immediately in the event the employee no longer has other Group Medical Insurance or the plan on which the City allowing the opt out has changed.
- F. The City reserves the right to withdraw its approval of an employee's opt out at any time for failure to provide satisfactory proof of alternative coverage and to change the benefit options available to all City employees.

29.06 ELIGIBILITY

An employee must meet the eligibility criteria provided by the plan documents in order to receive coverage set forth herein which may include waiting periods or exclusions. See HR for current benefits available based on your eligibility. In the event of a conflict between anything in this policy and the plan documents, the plan documents control.

29.07 RESERVATION OF RIGHTS

The City reserves its right to change or discontinue any benefits offered in their entirety, at any time in the City's sole discretion but will comply with all applicable laws, rules, and regulations. In the event this policy conflicts with then existing law, then existing law will control.